Contents lists available at ScienceDirect

ELSEVIER





Location Advantages, Governance Quality, Stock Market Development and Firm Characteristics as Antecedents of African M&As



Abongeh A. Tunyi^a, Collins G. Ntim^{b,*}

^a Accounting and Finance, Liverpool Hope University, Business School, Faculty of Arts and Humanities, Liverpool Hope University, Liverpool, UK
 ^b Financial Ethics and Governance Research Group, Department of Accountancy and Finance, University of Huddersfield Business School, University of Huddersfield, Huddersfield, UK

ARTICLE INFO

Article history: Received 11 May 2015 Received in revised form 29 January 2016 Accepted 30 January 2016 Available online 25 March 2016

Keywords: National governance quality Location advantages Stock market development Firm characteristics Mergers and acquisitions Africa

ABSTRACT

This study explores firm- and country-specific antecedents of African M&As. We use one of the largest datasets to-date, consisting of 1490 unique African firms (11,183 firm-year observations) from 1996 to 2012 from 15 African countries. Our results suggest that improvements in time-varying country-level factors, including location advantages (market size, human capital and efficiency opportunities), national governance quality, and stock market development are associated with an increase in the volume of M&A activity. Consistent with the resource-curse paradox, high resource endowments are not associated with increased levels of M&A. In support of the management inefficiency, but contrary to the traditional firm size hypotheses, African targets are generally characterised by declining stock returns and accounting profitability, but are more likely to be larger firms; suggesting the presence of information asymmetry concerns in their selection. Notwithstanding, we find evidence of heterogeneity across countries with inconsistent support for the established target prediction hypotheses. Overall, our analysis suggests that a model which combines firm- and country-specific factors better explains the observed variations in African M&A activity.

© 2016 The Authors. Published by Elsevier Inc. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

1. Introduction

Despite substantial research on foreign direct investment (FDI), in general and merger and acquisitions (M&As), in particular, little can be said about factors that moderate these phenomena in unique institutional contexts, such as the African continent. In acknowledgment, Shimizu et al. (2004) call for more theoretical development and empirical examination of determinants of cross-border M&As across different institutional contexts. This paper seeks to contribute to the international business strategy (IBS) literature by investigating the degree to which changes in merger and acquisition (M&A) activity across key African markets are explained by firm- and country-specific factors. Noteworthy is the fact that prior research on the subject has either focused on environmental or country-level moderators of FDI and M&A (e.g., Rossi and Volpin, 2004) or on firm-level factors affecting the selection of merger targets (Powell and Yawson, 2007), but rarely on both. Arguably, FDI through M&A is more likely to occur if both country- and firm-level conditions are satisfied and hence, there is a need for greater linkage of the two literatures. Drawing from a transaction cost economics theoretical perspective (Coase, 1937; Dunning, 1980), we examine the extent to which

http://dx.doi.org/10.1016/j.intman.2016.01.005

1075-4253/© 2016 The Authors. Published by Elsevier Inc. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

^{*} Corresponding author at: Financial Ethics and Governance Research Group, Department of Accountancy and Finance, University of Huddersfield Business School, University of Huddersfield, Queensgate Campus, Queensgate, Huddersfield HD1 3DH, UK.

E-mail addresses: tunyiat@hope.ac.uk (A.A. Tunyi), c.ntim@hud.ac.uk (C.G. Ntim).

location advantages (including; market size, resource endowments, human capital and efficiency opportunities), national governance quality and stock market development (i.e., country-level factors) explain observable variations in the volume of M&As across 15 African countries. Additionally, we explore the extent to which individual firm characteristics (i.e., firm-level factors) impact on the likelihood that specific firms within these countries will be acquired. Prior studies exploring country- and firmlevel antecedents of M&A activity are generally rare (Martynova and Renneboog, 2008; Gorton et al., 2009), and particularly acute in the case of Africa (e.g., Rossi and Volpin, 2004). We focus on Africa for a number of reasons. First, Africa has become an important economy on the world stage, but has been largely ignored in contemporary business research.

Fig. 1 shows the growth in the gross domestic product (GDP) between 1996 and 2012 for Sub-Sahara Africa, Organisation of Economic Development and Corporation (OECD) and the World. The average growth in GDP for Sub-Sahara Africa (4.57%) outstrips the average growth of OECD member countries (2.08%) and the World (2.83%) over the period. Despite this high level of growth, many African economies still lag behind the rest of the world in terms of development (McFerson, 2010).

M&A can, perhaps, bolster economic development in Africa by improving local business' capabilities, creating an international presence for local companies, providing access to global markets, providing financing for growth and generating the level of market competition necessary to spur efficiency in local companies (Stoian and Filippaios, 2008; Ferraz and Hamaguchi, 2002). Stoian and Filippaios (2008), for example, argue that foreign direct investment (FDI) through cross-border mergers can foster development in local economies through three channels. These include; the improvement of capabilities of local labour through training programmes and new management techniques, the building of backward and forward linkages with other domestic firms (e.g., supply chain partners) and through co-operation with local research institutions and universities (Stoian and Filippaios, 2008). Ferraz and Hamaguchi (2002) contend that cross-border M&As can spur the improvement of regulatory frameworks and macroeconomic management strategies (through the restructuring of public finances and the prevention of massive private sector bankruptcies) and the modernisation of local corporate governance, as well as, production capabilities. Cross-border M&As into developing economies (such as those in Africa) sometimes target underperforming sectors and undervalued assets, which might otherwise be wiped out from the economy (Ferraz and Hamaguchi, 2002). Further, prior research suggests that M&A plays a disciplinary role (Palepu, 1986; Powell and Yawson, 2007) by replacing inefficient management. This lends it the ability to encourage and safeguard investments in times of regulatory and market failures. Many cross-country studies on M&A in emerging markets (e.g., Alvarez and Marin, 2010) typically ignore key markets across the African continent, while others (e.g., Rossi and Volpin, 2004) focus only on major African economies, such as South Africa. This means little is known about M&A in countries, such as Ghana, Nigeria and Kenya, which are gaining economic traction on the world stage (McFerson, 2010). An understanding of the antecedents of African M&A is likely to be useful in informing economic and development policies in this region.

Second, despite these potential benefits of M&A for Africa, prior research has focused on exploring factors driving inward FDI, in general, but not M&As in particular, into the region. One reason for this is, perhaps, the historically small number of M&As into the region, which constrains reliable statistical analysis. Like most emerging markets, several African countries are characterised by relatively low M&A volume (Gomes et al., 2012). Rossi and Volpin (2004) attribute the relatively low M&A volume across the continent (between 1990 and 1999) to poor financial reporting quality and low investor protection. Prior evidence (e.g., Mitchell and Mulherin, 1996; Andrade et al., 2001; Andrade and Stafford, 2004; Martynova and Renneboog, 2008; Gorton et al., 2009) suggests that takeovers are most likely to occur in periods of economic recovery, coinciding with rapid credit expansion, burgeoning external capital markets and stock market booms. The evidence also suggests that takeover waves are frequently driven by industrial and technological shocks with regulatory changes, such as deregulation, acting as a catalyst (e.g., Martynova and Renneboog, 2008; Rhodes-Kropft and Viswanathan, 2004). Over the last two decades, several African countries have witnessed substantial economic growth (see Fig. 1), credit expansion, technological advancement and industrial expansion and thus greater M&A activity can be expected in this region (Moghaddam et al., 2014). Some evidence (e.g., Moghaddam et al., 2014) also suggests that multinational corporations (MNC) from emerging markets have begun to improve their competitive positions in the world stage by expanding through M&A. This highlights the importance of M&A in this region and the need for further research on the phenomenon.



Fig. 1. Chart of economic growth: Africa versus the world from 1996 to 2012.

Third, evidence from Transparency International and the World Bank Group, suggests that several African economies are characterised by poor national governance (as demonstrated by high levels of corruption, lack of democracy, low levels of transparency, and the presence of conflict and instability) and low levels of economic development. This experience is, for the most part, unique to the African continent. Prior research argues that this trend of poor national governance across the continent stifles economic growth through FDI by raising uncertainty and transaction costs – *"the grabbing hand hypothesis"* (Voyer and Beamish, 2004; Wei, 2000; Zhao et al., 2003). However, recent years have witnessed observable improvements in national governance, democracy, transparency and public accountability in a number of African countries (McFerson, 2009, 2010). It is interesting to therefore explore how the efforts to improve the quality of national governance across different African countries impacts on M&A activity in the region.

Our study makes the following contributions to the extant IBS literature. First, we explore country-level factors, including the Dunning (1980) location advantages (including market size, resource endowments, human capital and efficiency opportunities), national governance quality and stock market development that impact on the volume of M&A activity across different African countries. Second, we test the established M&A theory in a new context (i.e., out-of-sample analysis) by assessing the impact of firm-level characteristics on the level of M&A activity across different African countries. Here, our objective is to deduce whether the selection of takeover targets in this context is systematic and whether bidders select takeover targets to accomplish specific objectives in line with prior M&A theoretical predictions. We redevelop the traditional firm size hypothesis to take account of the unique characteristics of our new sample. In this regard, we explain why takeover likelihood and firm size for African firms have a relationship inconsistent with the prediction of the traditional firm size hypothesis. Prior takeover prediction studies (Palepu, 1986; Brar et al., 2009) employ firm-level characteristics as determinants of takeover likelihood. We extend this literature by showing that traditional prediction models augmented with country-level factors better explain firm takeover likelihood.

To the best of our knowledge, this is the first large scale (i.e., a sample of 1490 unique firms from 1996 to 2012 from 15 African countries, giving a total of 11,183 firm-year observations) study looking at the time-varying country- and firm-level determinants of M&A across the African continent. Our key findings are as follows. First, our evidence suggests that improvements in time-varying measures of location advantages (including market, human capital and efficiency opportunities), national governance quality (as measured by World Bank governance indices), and stock market development (stock market capitalisation, market volatility and number of traded stocks) are associated with increased M&A activity. Nonetheless, contrary to Dunning (1980), but consistent with the resource-curse paradox, the volume of M&A declines with natural resource endowments. Second, we find that takeover likelihood for African firms declines with a firm's stock market performance and accounting profitability, suggesting that target management inefficiency (Palepu, 1986) appears to drive the selection of suitable targets by domestic and international bidders. But contrary to the prior UK, US and EU evidence (Palepu, 1986, Powell and Yawson, 2007, Brar et al., 2009), takeover likelihood for African firms generally increases with firm size. This suggests that information asymmetry concerns, perhaps, shape the selection of potential targets by bidding firms. Our subsample results, however, show heterogeneity in firm-level antecedents of takeover likelihood across African countries. For example, we find that targets in Egypt and Morroco are more likely to be well-performing firms. Third, we find that our country-level factors can augment a traditional takeover prediction model to better explain the incidence of takeovers and the selection of takeover targets across Africa. This suggests that the likelihood that a firm will be acquired in a particular period is dependent not only on its individual characteristics but also on country-level and environmental conditions.

The rest of the paper is organised as follows. Section 2 discusses country-level antecedents of African M&As. Section 3 discusses firm-level antecedents of African M&As. Section 4 outlines the research design, data and method. Section 5 discusses the empirical results, whilst Section 6 presents the summary and conclusion.

2. Country-level antecedents of African M&As

Given the limited number of prior studies looking at antecedents of M&A – a particular mode of FDI – into developing countries, we draw from the general FDI literature to develop our theoretical postulations. Several studies in M&A prediction modelling implicitly assume that firm characteristics are the main time-varying factors that moderate their acquisition likelihood (Palepu, 1986; Powell, 2001). At a broader level, it can, perhaps, be argued that only when the surrounding environmental conditions are suitable, do individual firm characteristics moderate firm acquisition likelihood (Mitchell and Mulherin, 1996; Andrade et al., 2001; Rhodes-Kropft and Viswanathan, 2004; Andrade and Stafford, 2004; Martynova and Renneboog, 2008; Gorton et al., 2009). The relevance of suitable environmental conditions is, perhaps, even more critical in the context of developing countries, where issues such as the availability of factors of production, good governance and financial (stock market) development are particularly pertinent. To explore the country-level antecedents of M&As in Africa, we build on the well-established Eclectic paradigm or Ownership-Location-Internalisation (OLI) framework (Dunning, 1980, 1998) and consistent with Guisinger (2001), North (2005), Dunning (2009) and Stoian and Filippaios (2008), expand our model to incorporate institutional factors (such as national governance quality and stock market characteristics) which, perhaps, also moderate a country's attractiveness to M&A. We discuss these next.

2.1. Dunning's location advantages

The motivation for FDI has been subject to substantial research since the 1960s (Hymer, 1960a,b, 1968). While a significant amount of FDI activities tend to be in the form of M&A globally, this has not been the case in the African continent (OECD,

2006), explaining the sparse research on the subject. Indeed, Dikova and van Witteloostuijn (2007) note that 77% of FDI into developed countries is through M&A as compare to 33% for their developing counterparts. FDI provides an opportunity for growth for local firms, an alternative to diversifying within their home market or acquiring other local firms (Dunning, 1980, 1998). Dunning (1980) contends that some factors of production are company-specific (e.g., technology, knowledge, patents, trademarks, and brand names), while others (e.g., natural resources) are location-specific. The OLI paradigm (Dunning, 1980) stipulates that the decision to expand internationally makes economic sense when a firm "owns" certain assets (company-specific factors of production such as patents, knowledge and skill, and technology), which allows it to generate a competitive advantage over the challenges and costs of operating in an unfamiliar environment. The paradigm refers to these as "ownership advantages". In the current research, we focus on understanding the factors that make some firms within certain countries suitable M&A targets and hence, "ownership advantages", which generally pertain to the bidder's characteristics and are considered as "mobile assets" (Franco et al., 2010), are not central to our work. In our model, these are somewhat captured by the characteristics of targets selected by bidders, "Location advantages" in OLI framework explain why certain regions attract particular types of firms. Consistent with Dunning (1980, 1998), this "location advantages", perhaps, derive from the assets supplied by foreign markets, including new markets for products (market size), natural resources (resource endowments), low cost factors of production (efficiency-opportunities) and knowledge and skill (human capital). We discuss these issues and generate empirical predictions in the paragraphs that follow.

Prior research suggests that FDI is primarily driven by a market-seeking motive (Markusen, 1984; Brainard, 1997; Markusen and Venables, 1998, 2000). The goal of FDI per this motive is to exploit the host market by directly supplying it or its neighbours with goods and services (Franco et al., 2010). In this context, FDI (through M&A) allows firms to improve their competitiveness through a reduction in transportation costs or tariffs associated with exporting and by being more responsive to their customers (Brainard, 1997; Markusen and Venables, 2000; Franco et al., 2010). GDP figures from the African Development Bank and World Bank suggest that several African economies have witnessed tremendous market growth over the last decade. On average, year on year, African economies grew by 5.27% between 2000 and 2010 compared to the 2.77% global growth (see Fig. 1). The evidence, however, suggests significant variability within-country over time, as well as, across countries. Consistent with prior studies (Dunning, 1980; Maksimovic and Phillips, 2001; Harford, 2005), our first prediction is that market size (measured by GDP) explains variations in the volume of M&A across African markets. However, this might not be the case in the context of Africa as several MNCs expanding into developing markets have been shown to adopt a regional approach, where regional headquarters (RHQs) are sometimes used as vantage points for responding to customers across the entire region (Lasserre, 1996). As suggested by Luiz and Radebe (2012), the choice of RHQs in Africa is linked to advantages of agglomeration, economies of scale and the quality of the institutional environment at a later stage in this study.

Next, the OLI framework suggests that resource endowments (such as the presence of natural resources) might explain why firms choose to invest in some foreign countries (Dunning, 1980). Here, firms seek to acquire resources (i.e., raw materials), which are either unavailable or available at a higher cost in their home country through FDI. Inconsistent with Dunning (1980), prior research suggests a negative association between natural resource availability and FDI into developing economies (Sachs and Warner, 1995, 1997; Poelhekke and van der Ploeg, 2010; Asiedu and Lien, 2011). Asiedu and Lien (2011), for example, argue that such a counter-intuitive relation could persist as high natural resources lead to currency appreciation, making exports less competitive and, hence, crowding out investments in non-natural resource sectors. While initial exploration of natural resources require significant capital outlay, ongoing operations are not capital intensive, leading to a decline in FDI after the initial exploration phase (Asiedu and Lien, 2011). Poelhekke and van der Ploeg (2010) show that natural resources boost resourcerelated FDI, but crowds-out non-resource FDI to the extent that the net effect is a reduction in aggregate FDI into the country. Sachs and Warner (1995) also contend that natural resources (such as oil) deter FDI as resource-dependent economies are characterised by booms and busts, and because such economies are likely to be undiversified, this leads to significant volatility in exchange rates and added vulnerability to external shocks. This counter-intuitive position is consistent with the popular economic paradox - the Resource Curse - which explains the underperformance of resource-abundant countries, particularly those in Africa (Sachs and Warner, 1997). Further, even if resource endowments are important for FDI, they are likely to explain FDI in particular industries, such as oil & gas exploration, which are typically pursued through Joint Ventures and Greenfield investments as opposed to full-fledged M&As (OECD, 2006; Dikova and van Witteloostuijn, 2007). Following the literature, but contrary to the predictions of the OLI framework, our second prediction is that the incidence of M&As amongst African countries will decline with natural resource availability. We use "Total natural resources rents as a percentage of GDP" (compiled by the World Bank Group) as a measure of each country's resource endowments.

Third, the OLI framework predicts that the presence of knowledge and skilled labour (human capital) can attract FDI into some countries (Dunning, 1980). Pfeffermann and Madarassy (1992) note that technological advances have led to a shift in FDI towards more knowledge- and skill-intensive industries, making economies with high education levels more attractive FDI destinations. Noorbakhsh et al. (2001) argue that MNCs enhance their competitiveness by organising themselves functionally, shifting key activities (such as R&D, IT, customer services, accounting, training, distribution and the production of components) to countries best suited (in terms of skills and costs) for such activities. A number of studies have examined the relation between human capital and FDI in different contexts, with inconclusive results. Early studies by Root and Ahmed (1979); Schneider and Frey (1985) and Narula (1996) do not find a significant relation between human capital in developing countries and inward FDI in the pre-1990 period. A study by Noorbakhsh et al. (2001) covering the period up to 1994 finds human capital is one of the most important determinants of inward FDI into Africa, Asia and Latin America. Consistent with Noorbakhsh et al. (2001), we predict that M&A volume will increase with the level of human capital. We use "Total number of patent applications" (compiled by the

World Bank Group) as our main measure of human capital. Additionally, for robustness, we explore other measures of human capital including "Total number of trademark applications" and "Research and Development expenditure (as a percentage of GDP)".

Further, the OLI paradigm stipulates that opportunities to generate efficiencies (perhaps, through cost savings) may attract MNCs to expand to certain regions (Dunning, 1980). Local firms can service foreign markets through exports but this becomes unsustainable when transportation costs and tariff barriers are high (Coase, 1937). In situations where a foreign market is attractive enough, firms can improve their competitive position by substituting exports with local production. Indeed, several studies have empirically shown that efficiency opportunities such as low production costs attract FDI (Asiedu and Lien, 2011; Pfeffermann and Madarassy, 1992). Consistent with Dunning (1980) and Coase (1937), we anticipate that opportunities for low cost production in certain African countries can increase their attractiveness to FDI through M&A. Our fourth prediction is that the volume of M&A will increase with efficiency opportunities. We use the "Pump price of diesel fuel in USD" and "Average wage in USD" (compiled by the World Bank Group) as simple proxies for cost of production.

The final element of the OLI paradigm "Internalisation" explains why firms might prefer to engage in production abroad as opposed to subcontracting (through licensing or exporting) with foreign partners. The element contends that firms produce abroad due to "internalisation advantages", which derives from the propensity to earn higher rents on firm assets and/or achieve lower transaction costs by engaging in production abroad rather than subcontracting (through licensing or exporting) with third parties abroad (Dunning, 1980). As noted above, the current study nonetheless only focuses on the "where" question as we seek to explore what location and institutional factors make certain African countries more attractive M&A hotspots than their counterparts.

2.2. National governance quality

The second country-level variable in our model explores the extent to which the quality of the institutional environment impacts on the volume of M&A in Africa. The African Development Bank views governance as the manner by which government power is exercised to attain social and economic development (Boas, 1998). Data from international organisations, such as the World Bank Group and Transparency International, suggests that when compared to Western nations, African countries are generally characterised by poor governance. This is supported by the often relatively high levels of corruption, political instability, low regulatory quality, lack of accountability and general ineffectiveness of government institutions across several African countries (McFerson, 2009; Kaufmann et al., 2010). This lack of national good governance is not only a salient characteristic to resource-rich countries – the so called 'resource curse' – but also shared by resource-poor African countries (McFerson, 2009). Recent research, nonetheless, suggests that some countries within the continent have shown significant improvements in terms of governance quality. Based on evidence from the results of the Ibrahim Index of African Governance and Data from the Freedom House annual global surveys of political rights and civil liberties, McFerson (2010), for example, argues that as a continent, Africa has made progress in terms of national governance improvements over time – with Botswana, Ghana, Mauritius, South Africa and Tunisia, often cited as examples of African countries that have recently experienced improvements in the quality of their national governance.

In a cross-country study, Rossi and Volpin (2004) find that countries with better accounting standards and stronger shareholder protection have a higher volume of M&A. Their sample includes a limited number of African countries, such as Kenya, Nigeria, South Africa and Zimbabwe. Other studies show that country-level factors, such as macroeconomic stability, level of corruption, natural resource endowments and level of financial development also explain the cross-sectional differences in the volume of M&As (e.g., Vencatachellum and Wilson, 2013). Li et al. (2012) find that Rule of Law (an indicator of the quality of national governance) affects value creation in international strategic alliances into BRIC countries (Brazil, Russia, India and China), by moderating the foreign partners' willingness to share valuable knowledge assets. Another study by Karhunen and Ledyaeva (2012) also explore how corruption impacts on FDI choices, noting the choice can be modelled as a trade-off between the benefits of having a local partner and the costs associated with the existence of high levels of corruption. In general, poor national governance generates added political risk and acquirers will require equally high returns to justify their decision to invest in countries with relatively poor national governance credentials. National governance quality, therefore, should impact on M&A activity as more M&A, especially those involving cross-border bidders, are likely to be pursued in times of political stability, non-violence, low corruption, effective governments and high quality regulation (Rossi and Volpin, 2004; Martynova and Renneboog, 2008; Vencatachellum and Wilson, 2013). Consequently, our fifth prediction is that improvements in national governance quality across countries and over time will lead to an increase in M&A activity. As in prior studies, we measure the quality of National Governance and a country's institutional environment using time-varying measures of corruption, including Transparency International's Corruption Perception Index (CPI) and the World Bank's Control of Corruption index (CCI), as well as, measures of political and legal maturity, including the World Bank's Government Effectiveness index (GEI), Regulatory Quality Index (RQI) and Rule of Law Index (ROLI).

2.3. Stock market development

A key contribution of our study is that we focus on M&As distinguishing our study from the studies that look at FDI flows in Africa. The level of stock market development is perhaps critical to M&As, but not to other channels of FDI as M&A involving listed companies generally require active stock markets. Allen et al. (2012) document the poor state of the financial sectors of countries in sub-Saharan Africa, noting that development in this sector trails that of other developing nations. With the exception of South Africa, the level of financial development in the continent is, arguably, low. This is characterised by low access to debt and bond

markets, small size of the banking sector, low size of deposits and loans, high level of non-performing assets and capital inadequacy (Hearn et al., 2010; Andrianaivo and Yartey, 2010; Wang et al., 2015). Recent years have seen the rapid development of African Stock Exchanges the adoption of international accounting and auditing standards by several African countries (Ntim et al., 2007; Ntim et al., 2011; Ntim et al., 2012; Ntim, 2012). The successful completion of M&A deals requires the existence of functioning stock markets, with reliable stock valuations and sufficient firms for bidders to choose from (Ntim et al., 2015). These markets will also provide liquidity to investors, reduce the cost of information gathering through collective wisdom in the pricing of assets and provide opportunities for the risk sharing and diversification. In the context of M&A, the pricing of targets, acquisition of toeholds, payment through stock and opportunity for investors to liquidate their investment, all need fully functioning capital markets. Consequently, our sixth prediction is that improvements in stock market development will lead to an increase in M&A activity.

3. Firm-level antecedents of African M&A

The first part of our model explores country-level factors (including location advantages, governance quality and stock market development) that moderate the volume of M&A across African countries. In this section, we explore individual firm-level factors affecting the selection of particular firms as suitable targets. Arguably, FDI through M&A is more likely to occur if country and firm-level conditions are met. Two main theories explaining the incidence of M&A amongst firms, the neoclassical and managerial theory, have been propounded. The neoclassical (or shareholder) theory of mergers proposes that mergers are planned and executed by managers aiming to create, increase or maximise shareholder wealth (Manne, 1965). Mueller (1969) contends that managers seeking to maximise shareholder wealth will engage in mergers when it increases the firm's market power, enables the firm to achieve managerial or technological economies of scale, or when acquiring managers hold superior information about the target which is unavailable to the target's stakeholders. A review of the literature (e.g., Trautwein, 1990) suggests that managers can maximise the wealth of shareholders through M&A in four key ways summarised under the efficiency theory of mergers (managers aim to create synergies through M&A), monopoly theory of mergers (managers aim to bolster market power), raider theory of mergers (managers aim to acquire undervalued assets through M&A), and valuation theory of mergers (managers have superior information about a target's value compared to the stock market). These theories can be used to develop predictions on firm-level determinants of takeover likelihood amongst listed firms in the African context. Here, we build on a takeover model specification employed across several studies (e.g., Palepu, 1986; Powell, 1997; Powell and Yawson, 2007; Brar et al., 2009; Cremers et al., 2009). Palepu (1986) uses proxies of six propositions to build his takeover likelihood model. These six postulations include the inefficient management, firm undervaluation, growth-resource mismatch, firm undervaluation, price-earnings and industry disturbance. Other researchers, such as Ambrose and Megginson (1992); Powell (1997) and Brar et al. (2009) have suggested other drivers of takeover likelihood including real property, free cash flow and firm age. These postulations are discussed and adopted in the paragraphs that follow.

Prior research (Palepu, 1986; Morck et al., 1989) suggests that poorly performing managers or firms are more likely to be targeted in takeovers. Here, the objective of the bidder is to generate value by more efficiently managing the resources available to incumbent target managers with the neoclassical goal of maximising the wealth of shareholders. Management inefficiency is generally proxied by a firm's accounting and stock market performance measures, such as its abnormal stock return, return on assets (sales or equity) and operating profit margin (Agrawal and Jaffe, 2003). Prior studies find evidence that targets experience a decline in stock returns (Powell and Yawson, 2007) and profitability (Cremers et al., 2009) prior to their acquisition.

Some takeovers are motivated by the desire to acquire underpriced firms (Palepu, 1986). In this case, the bidder perceives the potential target as relatively undervalued by the market given the book value of its assets. A bidder with superior management ability can benefit from this market discrepancy by purchasing the assets (target) and unearthing its true value. This could be through divestments and reorganisation. A European study by Brar et al. (2009) which revealed that takeover targets have significantly higher earnings to price ratios as well as dividend yields, provides some empirical support for this postulation.

Palepu (1986) contends that a mismatch between a firm's growth levels and its available financial resources will lead to takeovers as bidders see opportunity to create synergies by correcting such an imbalance. In essence, low-growth high-resource firms as well as high-growth low-resource firms are most likely to receive takeover bids (Palepu, 1986; Powell, 2001). Low-growth high-resource firms, for example, are those with significant liquidity, strengthened by low debt levels, but lacking in suitable investment opportunities. Empirical support for the hypothesis is mixed. Whilst Palepu (1986) provides evidence to support this prediction, other studies, such as Espahbodi and Espahbodi (2003) and Powell (2004) do not find support for the hypothesis.

Firms with excess free cash flow are likely to be attractive takeover targets as the presence of excess free cash flow exacerbates the agency problem (Jensen, 1986; Lehn and Poulsen, 1989). In the presence of excess free cash flow, managers are likely to indulge in unprofitable projects or expropriate shareholder funds by securing managerial perquisites. More importantly, the presence of excess free cash flow within a target, potentially, reduces the bidders implicit cost of acquisition as these resources can be used to directly offset the bidder's acquisition costs. Consistent with this view, prior studies such as Powell (1997); Espahbodi and Espahbodi (2003) and Brar et al. (2009) show that UK, US and European targets (respectively) have significantly higher levels of free cash flow when compared to their respective bidders.

Bidders are more likely to be attracted to firms with substantial tangible assets (property, plants and equipment) in their total asset portfolio (Ambrose and Megginson, 1992). There are several reasons for this. First, the acquisition of tangible property, potentially, reduces the bidders cost of borrowing as these assets can act as collateral security (Powell and Yawson, 2007). Second, tangible property (as compared to intangibles) is easier to value. Hence, the presence of substantial tangible property reduces the

problem of information asymmetry faced by the bidder in valuing the target. Third, tangible assets which are not core to the bidder's line of business can easily be sold post-acquisition to further reduce the cost of acquisition. Intangibles such as brands, networks, goodwill, do not confer such advantages. In support of the real property hypothesis, prior studies including Ambrose and Megginson (1992); Powell (1997) and Espahbodi and Espahbodi (2003) find that takeover probability increases with the proportion of tangible assets in a firm's total asset portfolio.

Smaller firms are likely to face a higher takeover threat due to a lower cost of acquisition and the relative ease of post-merger restructuring (Palepu, 1986; Powell, 1997; Brar et al., 2009). Per this hypothesis, larger are firms are more expensive to acquire, more likely to resist acquisitions, more difficult to integrate/absorb and the pool of potential bidders for larger firms is smaller. This view, however, fails to take into account the dynamics associated with firm size. The problem of information asymmetry and its effect on the market mechanism – the market for lemons – has been discussed in prior research (Akerlof, 1970). The market for firms is, perhaps, not an exception to this problem which is likely to be more pervasive in the African context. Some researchers (e.g., Pettit and Singer, 1985) argue that, due to a lack of economies of scale in information production and distribution, smaller firms are inclined to produce and distribute less information about themselves, thus leading to a higher level of asymmetry between them and their stakeholders. This problem of comparatively higher information asymmetry in smaller firms is further exacerbated by the lack of significant analyst following in smaller firms (Eleswarapu et al., 2004). This suggests that, if bidders are cautious of purchasing 'lemons', they are likely to bid for low information asymmetry firms – those which produce and distribute large volumes of information about themselves and are followed by several analysts. Further, affordability and transaction costs are unlikely to be a major concern for foreign bidders of African targets given the substantial disparity in the market value of listed companies in developed and developing economies. The implication is that bidders for African targets will be, perhaps, more inclined to acquire larger than smaller targets on average.

Younger firms are more susceptible to acquisitions (Agarwal and Gort, 2002). Substantial research has been done in the firm life cycle literature which focuses on understanding the different stages in the life cycle of a typical firm (including industry entry, growth, decline and exit). This literature frequently attributes firm survival (age) to the ability of firms to learn actively or passively over time (Hopenhayn, 1992; Pakes and Ericson, 1998; Bhattacharjee et al., 2009). In line with the learning perspective, Bhattacharjee et al. (2009) contend that exit rates (due to the hazard of takeovers or bankruptcies) should decrease with age. Agarwal and Gort (2002) advanced the literature on firm age and survival by proposing that two key factors (learning-by-doing and firm endowments) define its probability of survival (and hence likelihood of industry exit). Agarwal and Gort (2002) contend that, over time, a firm gains knowledge about itself and its industry, which allows it to achieve cost reductions, product improvements, and develop new market techniques — learning-by-doing. In terms of endowments, Agarwal and Gort (2002) argue that firm endowments (i.e., a firm's inherent or natural suitability for profitability) are generally low when firms are born, but increase over time as firms invest in research and development. Older firms are therefore more endowed and more knowledgeable about themselves. The implication is that the probability of firm survival within an industry increases as firms grow older, learn about themselves and improve their endowments.

The country- and firm-level antecedents of M&A discussed above together with their selected proxies are presented in the Appendix (Table A1). The proxies adopted are in line with those used in prior studies (Palepu, 1986; Powell, 2004; Powell and Yawson, 2007; Brar et al., 2009 and Cremers et al., 2009). Our proxies for location advantages (market size, GDP; resource endowments, resource rent; human capital, patent applications; efficiency opportunities, pump fuel price and average wage), national governance (Worldwide Governance Indicators, Kaufmann et al., 2010) and stock market development have been widely used in prior research (McFerson, 2010; Vencatachellum and Wilson, 2013). We discuss our model in the next section.

4. Data and method

Our first model, examines the relation between time-varying country-level factors (including the quality of national governance, economic development and financial development) and the level of M&A activities (M&A volume) in African economies. To investigate the effect of our country-level time-variant factors on M&A activity, we collect data on M&A activities, involving listed companies across 15 African stock markets (including Botswana, Egypt, Ghana, Nigeria, Ivory Coast, Kenya, Mauritius, Namibia, Morocco, Tanzania, Tunisia, Uganda, South Africa, Zambia and Zimbabwe). We exclude some markets from our analysis (e.g., Cameroon, Sierra Leone, Libya and Sudan) as firm-level data for these markets is unavailable. Consistent with Rossi and Volpin (2004), we measure M&A volume (Volume) as the ratio of M&A bids to the number of listed companies in each country in each year. Our model specification is of the following general form.

$$Volume_{jt} = \alpha + \beta \left(Location_{jt} \right) + \gamma \left(Governance_{jt} \right) + \delta \left(MarketDev_{jt} \right) + \varepsilon$$
(1)

For robustness, we use several alternative proxies available in public domain to measure location advantages (Location), national governance quality (governance) and stock market development (MarketDev). As discussed in Section 2.1, we proxy location advantages as follows: (1) market size; GDP, and GDP growth, (2) resource endowments; and resource rent as a proportion of GDP (3) quality of human capital; and number of patent applications, and (4) efficiency opportunities; price of fuel and average wage. As noted in Section 2.2, we proxy the level of National Governance (Governance) using time-varying measures of corruption including Transparency International's Corruption Perception Index (CPI) and the World Bank's Government Effectiveness

index (GEI), Regulatory Quality Index (RQI) and Rule of Law Index (ROLI). As discussed in Section 2.3, to assess capital market development, we use stock market capitalisation deflated by GDP, stock market total value traded deflated by GDP, market volatility and the number of listed companies in each year deflated by total number of listed companies in the sample. We discuss these measures further and provide information about the source of this data in the appendix (Table A1).

In our second model, we adopt the standard takeover likelihood modelling methodology, which posits that the likelihood for a firm to be selected as a merger or takeover target is based on observable firm characteristics (Powell, 2004). Given that our study has a cross-country focus, we extend this model by suggesting that a firm's likelihood of being selected as a target is dependent on its observable characteristics, as well as the prevailing macro-environmental conditions (such as national governance, economic performance and financial development). Following prior studies (including Palepu, 1986; Powell, 2001; Cremers et al., 2009), we adopt a logit framework for computing firm acquisition likelihood. In this framework, we assume that a firm's likelihood of being the subject of a takeover bid in period t, is a function of its last observable characteristics (in period t - 1). The basic model is shown below.

$$Logit(P_{it}) = Ln\left(\frac{P_{it}}{1 - P_{it}}\right) = \beta \cdot Z_{it-1}$$
(2)

 P_{it} is the takeover likelihood of firm *i* at time *t* and X_{it-1} is a vector of characteristics (firm and country-level moderators of acquisition likelihood) for firm *i* at time t - 1. P_{it} can be computed as the inverse of the logit function – i.e., the logistic function – as shown below.

$$P_{it} = Logit^{-1}(\beta \cdot X_{it-1}) = \frac{1}{1 + e^{-\beta Z_{it-1}}}$$
(3)

In Eqs. (2) and (3), Z is a vector of firm-level and country-level characteristics given by the following.

$$Z_{it-1} = \beta_0 + \beta_1 X_{1it-1} + \beta_2 X_{2it-1} + \dots + \beta_k X_{kit-1} + \varepsilon_{it-1}$$
(4)

 β_0 is the intercept term in the logit regression and β_j (j = 1,..., k) represents the coefficients associated with the corresponding independent variables X_j (j = 1,..., k) for each firm or country. The dependent variable in our model takes the value of one if a firm is the subject of a takeover in a period and a value of zero otherwise. An observation is defined as a target in period *t* (and takes a value of P = 1) if it receives a bid in the current year. Otherwise, the observation takes a value of zero.

In additional analysis, we explore different lags in our model given the low volume of M&A activity and the inactive nature of the market for corporate control in the region. Given the perceived high information asymmetry between targets and bidders, we also consider that the decision to acquire a firm in a particular year may be related to its characteristics over a number of years (such as the last three years or the last five years). The modification of the lags used in standard prediction models allows us to explore the dynamics of takeover likelihood and changes in firm characteristics over time. We employ STATA v.13 in conducting our analyses. This allows us to perform panel regressions, adjusting our regression coefficients and standard errors for the effects of firm, year and country level clustering in our data.

To test our second model, we first obtain a comprehensive list of all listed African Companies, live and dead from the Thomson DataStream database. Each firm is identified by a unique DataStream code. We focus on the sample of all firms listed in the period 1996 to 2012 as very few M&A activity involving African targets is recorded prior to this period. Our sample consists of 1490 unique firms with DataStream codes. Unlike prior studies which utilise a match-sampling procedure (e.g., Palepu, 1986; Powell, 2001; Brar et al., 2009), we use a panel data set wherein each firm in our sample contributes an observation in every year between 1996 and 2012 until it exits the market through a bankruptcy, delisting or takeover. Notice that firms enter and exit the sample at different points over the sample period, and hence, over the 17-year period, the 1490 unique firms only contribute 11,183 firm-year observations. From the Thomson OneBanker database, we collect data on all announced takeover bids in every year between 1996 and 2012 where the target is a listed firm in Africa. We match both datasets using DataStream codes. Finally, we match our macro-level data (national governance quality, economic performance and financial development) to this dataset using a unique country-year identifier which we generate. We discuss our empirical results in the next section.

5. Empirical results and discussion

5.1. Trends in African M&A

First, we explore the incidence of M&A activity across Africa. Table 1 (Panel A) shows the distribution of listed firms, firm-year observations and M&A targets by country. There were over 1490 firms in Africa during the period 1996–2012 — a total of 11,183 firm-year observations. South Africa, Egypt, Nigeria and Morocco are the countries with the highest number of listed firms as well as the highest level of M&A activity. Even though Nigeria has more listed firms than Morocco, it has fewer M&A bids during the period. When organised by year as in Table 1 (Panel B), we find that, although, the number of M&A bids have steadily increased since 1996, the number of listed firms has increased at a faster pace leading to a slight decline in the proportion of listed firms that receive taker bids. Some of the early years (1996, 1997 and 2000) saw levels of M&A activity above 5%. In line with trends

Table 1

Sample distribution and characteristics.

Panel A: distribution of sample by country.								
Country	Firms	Obs	Targets	Country	Firms	Obs	Targets	
Botswana	17	120	0	Nigeria	106	653	17	
Egypt	217	1570	40	South Africa	795	6225	324	
Ghana	29	209	2	Tanzania	5	36	0	
Ivory Coast	29	192	0	Tunisia	53	362	3	
Kenya	54	387	3	Uganda	6	41	0	
Mauritius	41	296	2	Zambia	15	113	0	
Morocco	79	681	21	Zimbabwe	37	247	5	
Namibia	7	51	2	Total	1490	11,183	419	

Panel B: distribution of sample by year

Year	Firms	Targets	Target%	Year	Firms	Targets	Target%
1996	181	12	6.63%	2005	899	18	2.00%
1997	221	12	5.43%	2006	968	22	2.27%
1998	507	20	3.94%	2007	991	28	2.83%
1999	561	26	4.63%	2008	972	33	3.40%
2000	560	31	5.54%	2009	954	46	4.82%
2001	513	18	3.51%	2010	919	34	3.70%
2002	474	19	4.01%	2011	900	39	4.33%
2003	475	19	4.00%	2012	583	29	4.97%
2004	505	13	2.57%	Total	11,183	419	3.75%

Panel C: M&A in Africa by industry

Industry	Firms	Targets	Industry	Firms	Targets
	S	S		S	S
Aerospace & defense	3	0	Mining	785	52
Automobiles & parts	170	5	Oil & gas prodn & svcs	213	6
Chemicals	285	8	Personal goods	261	10
Construction & materials	723	34	Pharmaceuticals & biotech	189	9
Electronics & equipment	328	13	Real estate trusts & svcs	877	30
Financial services	2212	78	Retail	660	21
Food & beverage producers	1013	31	Software & computer svcs	443	18
Forestry & paper	83	2	Support svcs	425	16
Utilities	67	0	Technology hardware & equipment	146	4
Health care	108	2	Telecommunications	194	8
Household goods & construction	109	6	Tobacco	35	0
Industrials	1013	38	Travel & leisure	499	16
Media	224	11	Others	118	1

Notes: Panels A, B, and C show the distribution of listed firms (firm-years) and M&A targets in the sample grouped by country, year and industry, respectively. In Panel A, Firms refers to the number of firms listed on the country's main stock exchange, Obs refers to the number of firm-year observations for which data is available and Targets refers to the number of takeover (M&A) bids associated to these firm-year observations. In Panel B, Target% refers to the proportion of firms that receive takeover bids. In Panel C, Firms refer to the number of listed firms within each industry grouping per DataStream classification.

in UK and US, M&A peaked in 2000. There is a decline in M&A activity after 2004 coinciding with an increase in number of listed firms. There is no evidence of a significant decline in M&A activity during the 2007–2012 Global Financial Crises and Recession. Levels of M&A in 2012 are similar to those in 2000 with about 3.93% of listed African firms receiving takeover bids over the 2007–2012 period. This level of M&A activity is only slightly less than the 5.00% reported in developed economies such as the UK (Danbolt et al., forthcoming).

As evident in Panel C, M&A activity in Africa cluster by industry. The most active M&A industries include financial services (including banking and insurance), mining, industrials (including transportation), construction and materials and food & beverage producers. Other industries with moderate levels of activities include real estate investment trusts & services, retail (including food & drugs), software & computer services, support services and travel & leisure. There is limited or no M&A activity in industries such as aerospace & defence, automobiles and parts, forestry & paper, utilities (electricity, gas & water), tobacco, technology hardware & equipment and healthcare equipment & services.

5.2. Country-level moderators of M&A activities

Next, we empirically examine the relation between time-variant country-level variables and the volume of M&A activity as specified by Eq. (1). The results are shown in Table 2. The dependent variable in all our regression models (models 1 to 15) is M&A Volume defined as the proportion of listed firms in each country which receive takeover bids in the respective year. The independent variables in the model include proxies of location advantages, national governance quality and stock market

Table 2

Regression models for country-level determinants of M&A volume.

Panel A: M&A volume	and location advantag	ges				
	Model 1	Model 2	Model 3	Model 4	Model 5	Model 6
LnGDP	0.009***		0.010***	0.005***	0.010***	0.011***
GDPGrowth	(0.000)	-0.005	(0.000)	(0.000)	(0.000)	(0.000)
ResourceRent		(000)	-0.001^{***}			
Patent			(0.002)	0.004***		
FuelPrice				()	-0.014^{**}	
AverageWage					()	-0.014^{***} (0.004)
Constant	-0.196^{***} (0.000)	0.018*** (0.000)	-0.220^{***} (0.000)	-0.117^{**} (0.018)	-0.210^{***} (0.000)	- 0.252***
Observations	174	160	174	158	158	152
R-squared	0.129	0.001	0.168	0.189	0.176	0.308
Panel B: M&A volume	and national governar	nce quality				
	Model 7	Model 8	Model 9	Model 10	Model 11	Model 12
LnGDP	0.009*** (0.000)	0.009*** (0.000)	0.009*** (0.000)	0.009*** (0.000)	0.009*** (0.000)	0.009*** (0.000)
CPI	0.003*** (0.008)					
GEI		0.006** (0.028)				
VAI			0.006* (0.063)			
RQI				0.002 (0.631)		
ROLI					0.003 (0.427)	
CCI						0.006*
Constant	-0.216^{***}	-0.197^{***}	-0.208^{***}	-0.203^{***}	-0.206^{***}	-0.204^{***}
Observations	168	157	157	157	157	157
R-squared	0.176	0.169	0.170	0.153	0.155	0.170
Panel C: M&A volume	and stock market dev	elopment				
		Model 13		Model 14		Model 15
LnGDP		0.006***		0.005*		0.005**
SMCap		0.000**		(0.037)		(0.013)
MktVol		(0.020)		0.001* (0.085)		
FirmConc				(····-·)		0.844*** (0.000)
Constant		-0.138*** (0.006)		-0.118^{*} (0.074)		-0.097** (0.035)
Observations		160		160		174
R-squared		0.169		0.152		0.172

Notes: The dependent variable in the model is M&A volume defined as the proportion of listed firms within each country that receive a takeover bid in any particular year. Panel A shows the relation between M&A volume and location advantages (including, market size (LnGDP, GDPGrowth), human capital (patents), natural resources endowments (ResourceRent), efficiency opportunities (FuelPrice, AverageWage)).Panel B shows the relation between M&A volume and measures of national governance quality (including Transparency International's corruption perception index (CPI), the government effectiveness index (GEI), voice and accountability index (VAI), the regulatory quality index (RQI), the rule of law index (ROII) and the control of corruption index (CCI)). Panel C shows the relation between M&A volume and measures of stocks trading on the country's stock market as a proportion of total stocks traded in all African Stock Exchanges (FirmConc)). Variables are fully discussed in Appendix A. P-values are shown in parenthesis. ***, **, ** indicate significance at 1%, 5% and 10% levels, respectively.

development. In models 1 to 6 (Panel A), we directly test our prediction of a direct relation between location advantages and M&A volume following Dunning (1980). Generally, the results in Table 2 (Panel A) support our contention that M&A volume across the 15 countries is moderated by location advantages. Consistent with Dunning (1980), M&A volume increases with

market size (the market-seeking motive of FDI; Markusen and Venables, 2000). A change in Ln GDP of 1 unit increases the volume of M&A by 0.9%. In untabulated findings, our results remain robust when we adjust Economic performance by the size of the population (using GDP per Capita). The results also remain significant when we adjust for clustering by country using the Rogers (1993) methodology for computing clustered-robust standard errors. Consistent with Rossi and Volpin (2004), we do not find evidence that changes in the level of GDP growth from one year to another impacts on M&A volume. This suggests that while an increase in a country's GDP over time might stimulate M&A activity, changes in the magnitude of the growth rate from one year to another, in itself, does not have a direct impact on the volume of M&A activity. In our subsequent regressions, we consider Ln GDP as a control variable to evaluate the effects of other location advantages, national governance and stock market development on the volume of M&A activity.¹

Inconsistent with eclectic paradigm (Dunning, 1980) but in line with the resource-curse paradox and prior empirical research (Sachs and Warner, 1995, 1997; Asiedu and Lien, 2011), we find a negative relation between M&A volume and resource endowments. As accentuated by the resource curse paradox, we find that an increase in resource endowment (proxied by resource rent) by 1 unit decreases M&A volume by 0.1% (significant at the 1% level). This suggests that, within our African sample, the presence of natural resources does not directly improve the volume of M&A activity as suggested by the OLI framework (Dunning, 1980, 1998).

Consistent with Dunning (1980, 1998) and Noorbakhsh et al., 2001, the results suggests a positive relation between human capital, knowledge and skill (as proxied by patent applications) and the volume of M&A. An increase in human capital by 1 unit increases the volume of M&A by 0.4% (significant at the 1% level). Finally, the results show that measures of cost of production (fuel price and average wages) are negatively related to the volume of M&A. An increase in average wage and fuel price by 1 unit decreases M&A volume by 1.4% (significant at the 1 and 5% levels, respectively). This suggests that, consistent with (Dunning, 1980) the presence of opportunities to generate efficiency through low cost production, perhaps, partly explain variations in M&A volume across African countries. In summary, these results show that, consistent with Dunning (1980), market size, human capital and efficiency opportunities partly explain the incidence of African M&As. Contrary to Dunning (1980), however, we find that resource endowments do not explain the choice of M&A destination in the context of Africa. Indeed, consistent with the resource curse paradox and the studies that show a negative relation between resource endowments and FDI (Sachs and Warner, 1997; Sachs and Warner, 1995; Asiedu and Lien, 2011; Poelhekke and van der Ploeg, 2010), we also find that resource endowments have a negative impact on the volume of M&A activity across African countries.

In models 7 to 12 (Table 2, Panel B), we consider the relation between proxies of the quality of national governance in each year across the 15 countries and the volume of M&A activities in each country in the respective period. The results show a positive relation between the quality of national governance and the volume of M&A activity. The relation is statistically significant when national governance is proxied by CPI, GEI, VAI and CCI. For example, an increase in GEI by 1 unit is associated with a 0.6% increase in the volume of M&A activity (significant at the 5% level). These results suggest that, when we control for economic conditions across different countries and different years, improvements in the quality of national governance are associated with an increase in the volume of M&A. That is, countries with more effective governance systems (particularly, in terms of low corruption, high government effectiveness and high voice and accountability) are likely to see higher volumes of M&A.

In models 13 to 15 (Table 2, Panel C), we further evaluate the relation between measures of stock market development and volume of M&A activity. The results reveal that, when we control for differences in market size between countries and across time, the volume of M&A activity increases with stock market capitalisation, stock market volatility, the availability of traded stocks. An increase in the concentration of traded stocks by 1 unit is associated to 84.4% increase in the volume of M&A activity. That is, African countries with more developed stock markets in terms of the number of listed companies are likely to see higher M&A volumes than their counterparts. Overall, in support of our prediction, the results suggest that the level of stock market development positively impacts M&A activity.

The results in Table 2 suggest that the volume of M&A activity across our sample of African firms is moderated by measures of location advantages, national governance quality and stock market development. Arguably, several of the 15 country-level variables used in our analysis are interrelated. We therefore use principal component analysis to compress these 15 country-level variables into theoretically meaningful components. This will act as a robustness check on the findings in Table 2. We restrict our components to those with eigenvalues of at least 1. This allows us to generate 4 components (Comp1, Comp2, Comp3 and Comp4) with eigenvalues of 6.687, 3.126, 1.413 and 1.130, respectively. These 4 components cumulatively explain 82.38% of variation in our variables. Table 3 shows the loadings on each of the four components. For clarity we, exclude results where the magnitude of loadings is less than 0.30. The overall Kaiser–Meyer–Olkin (KMO) measure of sampling adequacy is 0.80 which is above the recommended minimum of 0.50 for principal component analysis (Cerny and Kaiser, 1977).

Table 3 (Panel A) shows that measures of national governance quality load highly on the first component (Comp1). The second component (Comp2) has a mix of measures of stock market development, market size and human capital. Measures of efficiency opportunities (FuelPrice and AverageWage) load highly on Comp3. Our proxy for natural resource endowments, as well as GDP growth loads highly on Comp4. In Panel B (Table 3), we regress M&A volume (as the dependent variable) on our 4 components (as independent variables). We find a positive and significant relation between M&A volume and Comp1 and 2, suggesting that national governance, stock market development and market size impact on takeovers takeovers. The results also show a negative but statistically insignificant relation between M&A volume and Comp3, suggesting that

¹ In unreported results, we find that our results do not materially change when we do not use Ln GDP as a control variable.

Table 3

Principal component analysis and regression results.

Factor	Proxy	Comp1	Comp2	Comp3	Comp4	Unexplnd	KMO
Location advantages	LnGDP		0.487			0.179	0.687
	GDPGrowth				0.839	0.187	0.407
	ResourceRent				0.403	0.371	0.738
	Patent		0.342			0.107	0.778
	FuelPrice			0.673		0.275	0.340
	AverageWage			0.610		0.269	0.545
National governance quality	CPI	0.336				0.111	0.818
	GEI	0.356				0.059	0.857
	VAI					0.292	0.724
	RQI	0.352				0.080	0.872
	ROLI	0.312				0.070	0.864
	CCI	0.340				0.071	0.810
Stock market development	SMCap		0.315			0.081	0.825
	MktVol					0.405	0.815
	FirmConc		0.367			0.086	0.879
Overall	Eigenvalue	6.687	3.126	1.413	1.130		
	Rho	0.446	0.208	0.094	0.075		
	КМО						0.080
Panel B: Regression results with	components						
		Comp1	Comp2	Comp3	Comp4	Constant	
	Coef.	0.004***	0.007***	-0.001	-0.001	0.015***	
	P-value	(0.000)	(0.000)	(0.587)	(0.352)	(0.000)	

Notes: The table shows results from principal component analysis (PCA) followed by regression analysis of identified components. We derive components in Panel A. The proxies for location advantages, national governance quality and stock market development are fully explained in Appendix A. We restrict our components to those with eigenvalues of at least 1. For clarity we only present results for magnitude of the loading when it is at least 0.3. Unexplnd refers to the proportion of variability in each proxy unexplained by the four components. KMO is the Kaiser–Meyer–Olkin measure of sampling adequacy. In Panel B, we regress the four components on M&A volume. P-values are shown in parenthesis. ***, **, * indicate significance at 1%, 5% and 10% levels, respectively.

efficiency opportunities increase the likelihood of M&As. Consistent with the finding that resource endowments do not stimulate M&A, we find a negative (though statistically insignificant) relation between M&A volume and component 4. We note however that much of the variability (37.12%) of the variability in our proxy for resource endowment is not captured by components 1 to 4. The adjusted R square of this regression model is 0.384.

5.3. Firm-level moderators of acquisition likelihood

Next, we investigate the time-varying firm-level factors that moderate firm acquisition likelihood. We start by looking at descriptive statistics for firm characteristics across our sample. In untabulated results,² we find that the mean return on asset (ROA) for African firms over the sample period is 5.6%. Nonetheless, there is significant heterogeneity amongst firms and countries (standard deviation of 9.4%) with countries such as Botswana, Tanzania and Namibia achieving mean return on assets of over 10%. The mean excess monthly return (AAR) as against the Morgan Stanley Capital International (MSCI) Emerging Markets Index is close to zero as expected. This suggests that on average firms in these African countries perform in line with the all other emerging markets. The results show that book to market (BTM) values for African firms are above 1 on average, with Egypt, Tunisia, Uganda and Zimbabwe, particularly, experiencing depressed market values. Such depressed market values could partly be explained by the political, economic and regulatory climate within these countries. We find significant differences between countries in terms of average firm size (SIZE), but these results cannot be given too much meaning due to the significant differences in the number of listed companies. For example, as shown in Table 1, Tanzania, Uganda and Namibia have less than 7 listed firms each while South Africa has over 795 firms. The average age (since incorporation) of firms in the panel sample is 7.8 years, with countries such as Botswana, Ghana, Tanzania and Uganda having several newly incorporated firms.

Table 4 presents results of univariate analysis (difference of means tests) comparing the financial characteristics of targets to non-targets. Panel A compares the firm (financial) characteristics of targets to non-targets for all firms in the sample. Panels B, C, D and E report the results obtained for South Africa, Egypt, Morocco and Nigeria, respectively.³

² Available upon request.

³ We do not present results for other countries in our sample as we do not have sufficient observations for robust independent statistical analysis.

Tabl	e 4
------	-----

Difference of Means.

Panel A: Juli sample		Targets	NTargets	Diff	(T stat)
NAR 0.007 0.003 0.004* (-1.52) FIM 3.859 3.896 -0.001 (-0.05) FIM 3.859 3.896 -0.001 (-0.05) IQ 0.130 0.123 0.005 (0.915) IQ 0.037 0.013 (0.559) FCF 0.007 0.015 -0.009 (0.237) SZE 14.166 13.816 0.311***********************************	Panel A: full sample				
ARR -0.007 -0.003 -0.0047 (-1.320) STM 3.859 3.896 -0.037 (-1.320) SCR 0.205 0.206 -0.001 (-0.051) LV 0.285 0.272 0.013 (0.559) LV 0.285 0.272 0.013 (0.123) STAR 0.290 0.290 0.000 (0.123) STAR 0.290 0.290 0.000 (0.123) STAR 0.912 7.722 2.190*** (3.175) STAR 0.917 -0.055 -0.002 (-0.877) ARR -0.007 -0.055 -0.002 (-0.871) ARR -0.007 -0.055 0.004 (0.913) IQ 0.155 0.231 0.004 (0.913) IQ 0.156 0.024 0.014 (0.912) SIZE 13.896 13.372 0.524 (-0.637) IQ 0.0289 0.23 0 (-0.012)	ROA	0.053	0.056	-0.003	(-0.521)
FIM 3.859 3.896 -0.037 (-1.307) SCR 0.0205 0.0268 -0.001 (-0.051) LQ 0.130 0.123 0.006 (0.915) LV 0.0285 0.272 0.013 (0.559) FCF 0.007 0.015 -0.000 (0.024) SIZE 14.166 13.816 0.351"************************************	AAR	-0.007	-0.003	-0.004*	(-1.857)
SCR 0.205 0.206 (-0.031) IQ 0.130 0.123 0.006 (0.955) LIV 0.285 0.272 0.013 (0.559) LIV 0.285 0.272 0.013 (0.559) TANG 0.290 0.230 0.000 (0.024) SIZE 14.166 13.816 0.351*** (0.024) SIZE 14.166 13.816 0.351*** (557) Partel E: South Africa - 0.005 -0.002 (-0.673) RAA -0.007 -0.005 -0.002 (-0.673) SIZE 0.136 0.128 0.004 (0.458) SIZE 0.328 0.29 0 (-0.071) SIZE 13.896 13.372 0.244 (0.912) SIZE 13.896 13.372 0.244 (-0.271) SIZE 13.896 1.372 0.246 (-0.267) SIZE 13.896 1.372 0.246 (-0.267)	BTM	3 859	3 896	-0.037	(-1300)
IQ 0.130 0.123 0.006 (.015) IEV 0.285 0.272 0.013 (.015) FCF 0.007 0.015 -0.009 (.025) SIZE 14.166 1.3.816 0.051" (.033) AGE 9.912 7.762 2.190"** (.557) Panel & South Africa (.023) (.015) RAR -0.007 -0.003 -0.001 (.023) SKR 0.2 0.166 (.0.66) (.0.63) LQ 0.136 0.128 0.004 (.0.18) LQ 0.136 0.29 0 (0.012) CF 0.008 0.011 -0.002 (0.37) LQ 0.136 0.289 0.29 0 (0.012) CF 0.008 0.011 -0.002 (0.37) LQ 0.135 0.289 0.024 (.0.21) SGR 0.028 0.011 -0.026 (0.37)	SGR	0.205	0.206	-0.001	(-0.053)
IV 0.228 0.272 0.013 (0.559) FCF 0.007 0.015 -0.009 (-1.525) TANG 0.200 0.230 0.000 (0.024) SIZE 14.166 13.816 0.351"*** (5.557) Panel E. South Africa (0.023) (0.023) RAR -0.007 -0.005 -0.002 (-0.873) AAR -0.007 -0.005 -0.002 (-0.873) SCR 0.2 0.166 0.004 (0.633) SCR 0.2 0.166 0.008 (0.663) LV 0.289 0.29 0 (-0.012) SCR 0.295 0.281 0.014 (0.663) LV 0.289 0.29 0 (-0.021) SIZE 13.896 1.3172 0.244*** (2.408) LV 0.285 0.281 0.014 (-0.127) TANG 0.295 0.281 0.0164 (-1.2657)		0.130	0.123	0.006	(0.915)
FCF 0.007 0.015 -0.009 (-1.52) TANG 0.220 0.230 0.000 (0024) SZE 14.166 13.816 0.351"" (3.75) AGE 9.912 7.762 2.190"" (5.57) Panel E. South Africa (0.55) 0.033 0.001 (0.23) AAR -0.007 -0.002 (-0.67)3 (0.63) (0.63) SGR 0.2 0.196 0.004 (0.68) LQ 0.135 0.128 0.004 (0.61) LV 0.289 0.29 0 (-0.012) CFF 0.008 0.011 -0.004 (0.912) SIZE 13.896 13.372 0.524"" (4.40) AGE 0.051 0.063 -0.014" (-0.52) SIZE 13.896 13.372 0.524"" (4.20) LV 0.264 0.217 0.014" (-0.52) SIZE 13.896 13.372 0.023"<	LEV	0.285	0.272	0.013	(0.559)
TANG 0.200 0.209 0.000 (.0024) SZE 14.166 13.816 0.351"" (.375) ACE 9.912 7.722 2.190"" (.375) ACE 9.912 7.722 2.190"" (.375) AR -0.007 -0.005 -0.002 (0.87) AAR -0.007 -0.005 0.008 (.063) SCR 0.2 0.196 0.008 (.063) LQ 0.136 0.128 0.008 (.063) LQ 0.136 0.128 0.008 (.063) LV 0.289 0.29 0 (0.012) TANG 0.255 0.281 0.014 (.0912) TANG 0.255 0.281 0.014 (.012) TANG 0.252 0.268 -0.014" (1.663) AAR -0.011 0.003 -0.014" (1.67) CR 0.013 -0.025 (1.474) LQ 0.129	FCF	0.007	0.015	-0.009	(-1525)
SZE 14.166 13.115 0.3.51*** (3.175) AGE 9.912 7.722 2.190*** (5.557) AGE South Africa (5.557) ROA 0.0055 0.053 0.0011 (0.237) ARA -0.007 -0.002 -0.002 (-0.873) BTM 3.38 3.274 0.106 (0.363) LQ 0.136 0.128 0.004 (0.863) LV 0.289 0.29 0 (-0.012) FCF 0.008 0.011 -0.002 (-0.371) SIZE 13.896 13.372 0.524*** (4.400) AGE 13.896 13.372 0.524*** (4.402) AGE 13.896 13.372 0.524*** (4.402) AGE 13.896 13.372 0.248 (-0.027) AGE 0.051 0.067 -0.016 (-1.633) AR 0.013 0.014* (-0.027) (-0.644)	TANG	0.290	0.290	0.000	(0.024)
Ace 9.912 7.722 2.190*** (2.507) Panel E: South Africa (5.557) RAA -0.007 -0.005 -0.002 (-0.873) AAR -0.007 -0.005 -0.002 (-0.873) SGR 0.2 0.196 0.004 (0.138) UQ 0.136 0.128 0.008 (0.963) LV 0.289 0.29 0 (-0.071) TANG 0.025 0.281 0.014 (0.912) SIZE 1.3896 1.372 0.514**** (2.408) SIZE 1.3896 1.372 0.514**** (2.408) AGE 0.051 0.067 -0.016 (-1.637) SIZE 1.3896 1.372 0.514**** (2.408) AGE 0.051 0.067 -0.016 (-1.637) RAR -0.011 0.003 -0.014** (-2.067) SIT 0.054 0.162 -0.027 (-0.464) LQ	SIZE	14 166	13.816	0.351***	(3.175)
Number Numer Numer Numer <td>AGE</td> <td>9 912</td> <td>7 722</td> <td>2 190***</td> <td>(5.557)</td>	AGE	9 912	7 722	2 190***	(5.557)
PanelE South Africa 0.005 0.001 (0.237) AAR -0.007 -0.005 -0.002 (-0.873) SCR 0.2 0.196 0.004 (0.188) UQ 0.136 0.128 0.0068 (0.063) LV 0.289 0.29 0 (-0.012) CF 0.0068 0.011 -0.002 (-0.371) TANG 0.295 0.281 0.014 (0.912) SIZE 13.896 13.372 0.524*** (2.408) Panel C: Exput (2.408) (-0.051) (-0.637) AAR -0.011 0.003 -0.014** (-2.057) SIM 5.252 5.698 -0.044 (-0.396) SCR 0.135 0.162 -0.027 (-0.646) UQ 0.129 0.154 -0.026 (-1.148) LEV 0.264 2.077 0.021* (-5.52) SCR 0.135 1.129** (1.367) (1.29*)		01012		21100	(0.007)
ROA 0.055 0.003 0.001 (0.237) BTM 3.38 3.274 0.106 (0.363) BTM 3.38 3.274 0.106 (0.363) LQ 0.136 0.128 0.008 (0.663) LV 0.289 0.29 0 (-0.012) FCF 0.008 0.011 -0.002 (-0.371) TNAG 0.285 0.281 0.004 (0.917) ACE 1.0863 3.738 1.126*** (4.420) ACE 1.0863 9.738 0.126*** (2.468) Panel C Egypt 7 0.016 (-1.653) RAA 0.051 0.067 -0.016 (-1.663) SGR 0.135 0.162 -0.027 (-0.464) LQ 0.129 0.154 -0.026 (-1.148) LV 0.264 0.217 0.044* (-1.527) ACE 8.966 7.757 1.229* (1.763) CAC <t< td=""><td>Panel B: South Africa</td><td></td><td></td><td></td><td></td></t<>	Panel B: South Africa				
AAR -0.07 -0.005 -0.002 (-0.873) BTM 338 3.274 0.106 (0.363) SGR 0.2 0.196 0.004 (0.188) LQ 0.136 0.128 0.008 (0.653) LFV 0.289 0.29 0 (-0.012) FAR 0.005 0.014 (0.912) SIZE 13.866 13.372 0.524*** (2.408) Panel C. Egypt (2.408) (-1.653) AAR -0.011 0.0067 -0.016 (-1.653) AAR -0.011 0.0067 -0.016 (-1.653) AAR -0.011 0.003 -0.014** (-2.057) BTM 5.252 5.658 -0.026 (-1.481) LQ 0.129 0.154 -0.026 (-1.481) LQ 0.264 0.217 0.047 (0.522) SGR 0.068 0.001 0.003 (-0.522) SIAC 3.58	ROA	0.055	0.053	0.001	(0.237)
BTM 3.38 3.274 0.106 (0.36) SGR 0.2 0.196 0.008 (0.188) LQ 0.136 0.128 0.008 (0.963) LEV 0.289 0.29 0 (-0.012) FGE 0.008 0.011 -0.002 (-0.371) TANC 0.295 0.281 0.014 (0.912) SIZE 13.896 13.372 0.524**** (4.420) AGE 1.0863 9.738 1.126*** (2.408) Panel C: Egypt . . (-1.663) AGR -0.011 0.003 -0.014** (-2.067) DIM 5.252 5.698 -0.446 (-0.396) SGR 0.135 0.162 -0.027 (-0.464) LQ 0.129 0.154 -0.026 (-1.148) LQ 0.281 0.302 -0.021 (-0.522) SIZE 14.76 13.58 1.179*** (3.976) LSW	AAR	-0.007	-0.005	-0.002	(-0.873)
SCR 0.2 0.196 0.004 (0.188) LQ 0.136 0.128 0.008 (0.93) LEV 0.289 0.29 0 (-0.012) TANC 0.295 0.281 0.014 (0.912) SIZE 13.896 13.772 0.524*** (4.40) ACE 10.863 9.738 1.126*** (2.408) Panel C. Egypt -0.016 (-1.653) ACR -0.011 0.003 -0.014** (-2.067) BTM 5.252 5.698 -0.0464 (-0.396) LV 0.264 0.217 -0.047 (0.592) FCF -0.001 0.033 -0.026 (-1.184) LV 0.264 0.217 -0.047 (0.522) SIZE 14.76 13.58 1.179*** (3.376) AGE 9.996 7.757 1.229* (1.63) Panel D: Moraco 0.023 -0.035 (0.624) LV<	BTM	3.38	3.274	0.106	(0.363)
LQ 0.136 0.128 0.008 (0.963) LFV 0.289 0.29 0 (-0.012) FCF 0.008 0.011 -0.002 (-0.371) SIZE 13.896 13.372 0.524*** (2.408) SIZE 13.896 13.372 0.524*** (2.408) Panel C. Egypt (2.408) (2.408) (2.408) Panel C. Egypt -0.016* (-1.663) RAR -0.011 0.003 -0.014** (-2.067) SGR 0.135 0.162 -0.027 (-0.464) LQ 0.129 0.154 -0.026 (-1.148) LV 0.264 0.217 0.047 (0.552) SIZE 1.476 13.58 1.179*** (3.976) AGE 0.281 0.302 -0.021 (-0.522) SIZE 1.476 3.584 1.79** (3.976) AGE 0.048 0.062 -0.015 (-1.326)	SGR	0.2	0.196	0.004	(0.188)
LEV 0.289 0.29 0 (-0.012) FF 0.008 0.011 -0.002 (-0.371) TANG 0.295 0.281 0.014 (0.912) SIZE 13.896 13.372 0.524*** (2.408) Panel C. Egypt (2.408) (-0.016) (-1.663) AAR -0.011 0.003 -0.014** (-2.067) BTM 5.252 5.698 -0.446 (-0.396) SGR 0.135 0.162 -0.027 (-0.464) LIV 0.264 0.217 -0.044 (-1.874) LIV 0.264 0.217 -0.047 (-0.592) SIZE -0.011 0.033 -0.034* (-1.874) TANG 0.281 0.302 -0.021 (-0.522) SIZE 14.76 1.58 1.179*** (3.976) AGE - 0.013 -0.024 (-0.915) AGE 14.76 1.58 1.179**** <	LIQ	0.136	0.128	0.008	(0.963)
FCF 0.008 0.011 -0.002 (-0.371) TANC 0.295 0.281 0.014 (0.912) SIZE 13.896 13.372 0.524*** (4.40) AGE 10.8063 9.738 1.126*** (2.408) Panel C: Egypt (-0.016 (-1.663) AAR -0.011 0.067 -0.014** (-2.067) BTM 5.252 5.698 -0.0446 (-0.366) IQ 0.129 0.154 -0.026 (-1.148) IQ 0.264 0.217 0.047 (0.592) FCF -0.001 0.033 -0.034** (-1.874) TANC 0.281 0.302 -0.021 (-0.522) SIZE 14.76 15.58 1.179**** (3.976) AGE 8.986 7.757 1.229* (1.763) Panel D: Morecco -0.015 (-1.326) AAR 0.008 0.001 0.007 (1.359)	LEV	0.289	0.29	0	(-0.012)
TANG 0.295 0.281 0.014 (0.912) SIZE 13.896 13.372 0.524*** (4.420) ACE 10.863 9.738 1.126*** (2.408) Panel C. Egypt (-1.663) (-2.067) RA -0.011 0.003 -0.014** (-2.067) BTM 5.252 5.698 -0.446 (-0.396) SGR 0.135 0.162 -0.027 (-0.464) LQ 0.129 0.154 -0.026 (-1.187) LEV 0.264 0.217 0.047 (0.592) SIZE 14.76 13.58 1.179*** (3.976) ACE 3.986 7.757 1.229* (7.53) Panel D: Morocco (-1.335) (-1.53) (-1.326) AAR 0.008 0.001 0.007 (1.359) SGR 0.166 0.129 0.036 (0.565) LQ 0.095 0.270 0.025 (0.254)	FCF	0.008	0.011	-0.002	(-0.371)
SIZE 13.896 13.372 0.524*** (4.420) AGE 10.863 9.738 1.126*** (2.408) AGE 0.051 0.067 -0.016 (-1.663) AAR -0.011 0.003 -0.014** (-2.067) BTM 5.252 5.698 -0.446 (-0.396) SGR 0.135 0.162 -0.027 (-0.464) LQ 0.129 0.154 -0.026 (-1.148) LEV 0.264 0.217 0.047 (0.592) PARC 0.033 -0.034* (-1.874) TANC 0.281 0.302 -0.021 (-0.522) SIZE 14.76 13.58 1.179*** (3.976) AGE 8.986 7.757 1.229 (1.763) Panel D: Moraco (-1.326) 0.007 (1.359) BTM 4.578 3.841 0.737 (0.633) SGR 0.166 0.129 0.036 (0.254) LQ	TANG	0.295	0.281	0.014	(0.912)
ACE 10.863 9.738 1.126*** (2.408) Panel C: Eggnt	SIZE	13.896	13.372	0.524***	(4.420)
Panel C: Egypt ROA 0.051 0.067 -0.016 (-1.663) JAR -0.011 0.003 -0.014** (-2.067) BTM 5.252 5.698 -0.446 (-0.396) SCR 0.135 0.162 -0.027 (-0.464) LQ 0.129 0.154 -0.026 (-1.187) LEV 0.264 0.217 0.047 (0.592) SIZE 1.476 1.3.58 1.179*** (.3.76) SIZE 1.4.76 1.3.58 1.179*** (.3.76) AGE 8.986 7.757 1.229* (.1.763) Panel D: Moreco (.1.359) (.1.737) (.633) SCR 0.166 0.129 0.036 (.0.241) LEV 0.295 0.270 0.025 (.0.241) LEV 0.295 0.270 0.025 (.0.241) LEV 0.295 0.270 0.025 (.0.241) LEV 0.295 0.270	AGE	10.863	9.738	1.126***	(2.408)
R0A 0.051 0.067 -0.016 (-1.663) AAR -0.011 0.003 -0.014** (-2.067) BTM 5.252 5.698 -0.446 (-0.396) SCR 0.135 0.162 -0.027 (-0.464) LQ 0.129 0.154 -0.026 (-1.163) LEV 0.264 0.217 0.047 (0.592) FCF -0.001 0.033 -0.034* (-1.874) TANC 0.281 0.302 -0.021 (-0.522) SIZE 14.76 13.58 1.179*** (3.976) ACE 8.986 7.757 1.22* (1.73) Panel D: Morocco (-1.326) (-1.326) AAR 0.008 0.001 0.007 (1.359) BTM 4.578 3.841 0.737 (0.633) GCR 0.166 0.129 0.036 (0.565) LQ 0.094 0.022 0.024 (-0.241) LV	Danal C. Equat				
NOA UGD1 UGD7 -OU16 (-1.683) AAR -0011 0.003 -0.014** (-2.067) BTM 5.252 5.698 -0.046 (-0.396) SGR 0.135 0.162 -0.027 (-0.464) LQ 0.129 0.154 -0.026 (-1.148) LEV 0.264 0.217 0.047 (0.592) FCF -0.001 0.033 -0.021* (-0.522) SIZE 14.76 13.58 1.179*** (3.976) AGE 8.986 7.757 1.229* (1.763) Panel D: Moreco (-1.336) (-1.336) AAR 0.0048 0.062 -0.015 (-1.326) AAR 0.0048 0.002 -0.008 (0.555) LQ 0.094 0.102 -0.008 (-0.54) SCR 0.166 0.129 0.035 (0.254) LV 0.295 0.270 0.025 (0.254) LV </td <td>Punel C: Egypt</td> <td>0.051</td> <td>0.067</td> <td>0.016</td> <td>(1.002)</td>	Punel C: Egypt	0.051	0.067	0.016	(1.002)
AAK - 0.011 0.003 - 0.014 (= 2.00) BTM 5.252 5.698 -0.446 (= 0.396) SGR 0.135 0.162 -0.027 (= 0.444) LIQ 0.129 0.154 -0.026 (= -1.148) LEV 0.264 0.217 0.047 (0.592) FCF -0.001 0.033 -0.034* (= 1.874) TANC 0.221 0.302 -0.021 (= 0.522) SIZE 14.76 13.58 1.179*** (3.976) AGE 8.986 7.777 1.229* (1.763) Panel D: Morocco R 0.062 -0.015 (-1.326) AR 0.008 0.001 0.007 (1.539) BTM 4.578 3.841 0.777 (0.633) GCR 0.166 0.129 0.036 (0.565) LIQ 0.094 0.102 -0.008 (-0.721) LEV 0.295 0.270 0.025 (0.254)	AAD AAD	0.051	0.007	-0.016	(-1.003)
DIM 3.2.32 3.038 U.440 (-0.390) SGR 0.135 0.162 -0.027 (-0.464) LQ 0.129 0.154 -0.026 (-1.148) LEV 0.264 0.217 0.047 (0.592) FCF -0.001 0.033 -0.034* (-1.874) TANG 0.281 0.302 -0.021 (-0.522) SIZE 14.76 13.58 1.179*** (3.976) AGE 8.986 7.757 1.229* (1.763) Panel D: Morocco (-1.326) (AAR (0.08 (0.001) 0.007 (1.359) FTM 4.578 3.841 0.737 (0.633) (G.555) IQ 0.094 0.102 -0.008 (-0.241) LEV 0.295 0.270 0.025 (0.254) IQ 0.094 0.021 -0.008 (-0.640) SGR 0.166 0.129 0.035 (0.520) TANG <t< td=""><td></td><td>-0.011</td><td>0.003</td><td>-0.014</td><td>(-2.007)</td></t<>		-0.011	0.003	-0.014	(-2.007)
SGR 0.133 0.162 0.027 (-0.404) LQ 0.129 0.154 -0.026 (-1.149) LEV 0.264 0.217 0.047 (0.592) FCF -0.001 0.033 -0.034* (-1.874) JARG 0.281 0.302 -0.021 (-0.522) SIZE 14.76 13.58 1.179*** (3.976) AGE 8.986 7.757 1.229* (1.763) Panel D: Morocco R 0.008 0.001 0.007 (1.359) BTM 4.578 3.841 0.737 (0.633) SGR 0.166 0.129 0.036 (0.556) LQ 0.094 0.102 -0.008 (-0.241) LEV 0.295 0.270 0.025 (0.254) CF 0.03 0.028 -0.024 (-0.970) TANG 0.178 0.211 -0.033 (-0.640) SIZE 15.442 14.636 0.806* (SCR	0.125	0.162	-0.448	(-0.590)
LQ 0.129 0.134 -0.026 (-1.148) LEV 0.264 0.217 0.047 (0.592) FCF -0.001 0.033 -0.034* (-1.874) TANG 0.281 0.302 -0.021* (-0.522) SIZE 14.76 13.58 1.179*** (3.976) ACE 8.986 7.757 1.229* (1.763) Panel D: Morocco (-1.326) (-1.326) AAR 0.008 0.001 0.007 (1.359) BTM 4.578 3.841 0.737 (0.633) SGR 0.166 0.129 0.036 (0.565) LQ 0.094 0.102 -0.008 (-0.241) LEV 0.295 0.270 0.025 (0.254) FCF 0.003 0.028 -0.024 (-0.970) TANG 0.178 0.211 -0.033 (-0.640) SIZE 15.442 14.636 0.806* (1.895)	SGR	0.135	0.162	-0.027	(-0.464)
LEV 0.264 0.217 0.047 (0.392) FCF -0.001 0.033 -0.034* (-1.874) TANG 0.281 0.302 -0.021 (-0.522) SIZE 14.76 13.58 1.179*** (3.976) AGE 8.986 7.757 1.229* (1.763) Panel D: Morocco K K K K ROA 0.048 0.062 -0.015 (-1.326) AAR 0.008 0.001 0.007 (1.359) BTM 4.578 3.841 0.737 (0.633) SGR 0.166 0.129 0.036 (0.254) LEV 0.295 0.270 0.025 (0.254) LEV 0.295 0.270 0.025 (0.254) CF 0.003 0.028 -0.024 (-0.970) TANG 0.178 0.211 -0.033 (-0.640) SIZE 15.442 14.636 0.806* (1.895)	LIQ	0.129	0.154	-0.026	(-1.148)
FCF -0.001 0.033 -0.034 (-1.374) TANG 0.281 0.302 -0.021 (-0.522) SIZE 14.76 13.58 1.179*** (3.976) ACE 8.986 7.757 1.229* (1.763) Panel D: Morocco - (-1.326) AAR 0.048 0.062 -0.015 (-1.326) AAR 0.008 0.001 0.007 (1.359) BTM 4.578 3.841 0.737 (0.633) SGR 0.166 0.129 0.036 (0.024) LQ 0.094 0.102 -0.008 (-0.241) LEV 0.295 0.270 0.025 (0.254) FCF 0.003 0.028 -0.024 (-0.970) TANG 0.178 0.211 -0.033 (-0.640) SIZE 15.442 14.636 0.806* (1.895) AGE 8.914 8.202 0.712 (0.520) Panel E:	LEV	0.264	0.217	0.047	(0.592)
IANG 0.281 0.302 -0.021 (-0.322) AGE 0.281 0.302 -0.021 (-0.322) AGE 8.986 7.757 1.229* (1.763) Panel D: Morocco (-0.132) (-0.132) ROA 0.048 0.062 -0.015 (-1.326) AAR 0.008 0.001 0.007 (1.359) BTM 4.578 3.841 0.737 (0.633) SGR 0.166 0.129 0.036 (0.565) LQ 0.094 0.102 -0.008 (-0.241) LV 0.295 0.270 0.025 (0.254) FCF 0.003 0.028 -0.024 (-0.970) TANG 0.178 0.211 -0.033 (-0.640) SIZE 15.442 14.636 0.806* (1.895) AGE 8.914 8.202 0.712 (0.520) Panel E: Nigeria -0.014 (-0.548) AAR <td< td=""><td>FCF</td><td>-0.001</td><td>0.033</td><td>-0.034</td><td>(-1.874)</td></td<>	FCF	-0.001	0.033	-0.034	(-1.874)
SLE 14.70 13.36 1.179 (3.370) AGE 8.986 7.757 1.229* (1.763) Panel D: Morocco ROA 0.048 0.062 -0.015 (-1.326) AAR 0.008 0.001 0.007 (1.359) BTM 4.578 3.841 0.737 (0.633) SGR 0.166 0.129 0.036 (0.055) LQ 0.094 0.102 -0.008 (-0.241) LEV 0.295 0.270 0.025 (0.254) FCF 0.003 0.028 -0.024 (-0.970) TANG 0.178 0.211 -0.033 (-0.640) SIZE 15.442 14.636 0.806* (1.895) AGE 8.914 8.202 0.712 (0.520) Panel E: Nigeria -0.017 -0.006 -0.011 (-0.912) BTM 11.911 5.759 6.152** (2.262) SGR 0.213 0.297 -0.084	LANG	0.281	0.302	- 0.021	(-0.522)
AGE 5.560 7.57 1.229 (1.75) Panel D: Morocco	SIZE	14.70	15.38	1.179	(3.970)
Panel D: Morocco ROA 0.048 0.062 -0.015 (-1.326) AAR 0.008 0.001 0.007 (1.359) BTM 4.578 3.841 0.737 (0.633) SGR 0.166 0.129 0.036 (0.565) LQ 0.094 0.102 -0.008 (-0.241) LEV 0.295 0.270 0.025 (0.254) TANG 0.178 0.211 -0.033 (-0.640) SIZE 15.442 14.636 0.806* (1.895) AGE 0.32 0.045 -0.014 (-0.548) AAR -0.017 -0.006 -0.011 (-0.912) Panel E: Nigeria 11.911 5.759 6.152** (2.262) SCR 0.213 0.297 -0.084 (-0.912) PTM 11.911 5.759 6.152** (2.262) SCR 0.213 0.297 -0.084 (-0.908) LQ 0.075 0.099	AGE	0.900	1.137	1.229	(1.705)
ROA 0.048 0.062 -0.015 (-1.326) AAR 0.008 0.001 0.007 (1.359) BTM 4.578 3.841 0.737 (0.633) GCR 0.166 0.129 0.036 (0.565) LQ 0.094 0.102 -0.008 (-0.241) LEV 0.295 0.270 0.025 (0.254) FCF 0.003 0.028 -0.024 (-0.970) TANG 0.178 0.211 -0.033 (-0.640) SIZE 15.442 14.636 0.806* (1.895) AGE 8.914 8.202 0.712 (0.520) Panel E: Nigeria	Panel D: Morocco				
AAR0.0080.0010.007(1.359)BTM4.5783.8410.737(0.633)SCR0.1660.1290.036(0.565)LQ0.0940.102-0008(-0.241)LEV0.2950.2700.025(0.254)FCF0.0030.028-0.024(-0.970)TANG0.1780.211-0.033(-0.640)SZE15.44214.6360.806*(1.895)AGE8.9148.2020.712(0.520)Panel E: NigeriaROA0.0320.045-0.014(-0.548)AAR-0.017-0.006-0.011(-0.912)BTM11.9115.7596.152**(2.262)SGR0.2130.297-0.084(-0.702)LQ0.0750.099-0.024(-0.969)LEV0.1990.1770.022(0.301)FCF-0.0260.006-0.032(-0.803)FCF0.0260.006-0.032(-0.803)LEV0.1990.1770.022(0.301)FCF-0.0260.006-0.032(-0.803)FCF0.0260.006-0.032(-0.803)FCF0.0260.006-0.032(-0.803)FCF0.0260.006-0.032(-0.803)FCF0.0260.006-0.032(-0.803)FCF0.0260.006-0.032(-0.803)FCF0.3140.3130.001	ROA	0.048	0.062	-0.015	(-1.326)
BTM 4.578 3.841 0.737 (0.63) SCR 0.166 0.129 0.036 (0.565) LIQ 0.094 0.102 -0.008 (-0.241) LEV 0.295 0.270 0.025 (0.254) FCF 0.003 0.028 -0.024 (-0.970) TANG 0.178 0.211 -0.033 (-0.640) SIZE 15.442 14.636 0.806* (1.895) ACE 8.914 8.202 0.712 (0.520) Panel E: Nigeria RAR -0.017 -0.006 -0.014 (-0.948) AAR -0.017 -0.006 -0.011 (-0.912) BTM 11.911 5.759 6.152** (2.262) IQ 0.075 0.099 -0.024 (-0.909) (-0.702) [LQ (0.301) (-0.911) (-0.926) [LQ (-0.702) [LQ (0.301) (-0.969) [LV < 0.199	AAR	0.008	0.001	0.007	(1.359)
SGR 0.166 0.129 0.036 (0.56) LIQ 0.094 0.102 -0.008 (-0.241) LEV 0.295 0.270 0.025 (0.254) FCF 0.003 0.028 -0.024 (-0.970) TANG 0.178 0.211 -0.033 (-0.640) SIZE 15.442 14.636 0.806* (1.895) AGE 8.914 8.202 0.712 (0.520) Panel E: Nigeria RAA -0.017 -0.006 -0.011 (-0.912) BTM 11.911 5.759 6.152** (2.262) SCR 0.213 0.297 -0.084 (-0.702) LQ 0.075 0.099 -0.024 (-0.969) LEV 0.199 0.177 0.022 (0.301) FCF -0.026 0.006 -0.032 (-0.803) 1.40*** (2.80) SIZE 17.539 16.569 0.970*** (3.280)	BTM	4.578	3.841	0.737	(0.633)
LIQ 0.094 0.102 -0.008 (-0.241) LEV 0.295 0.270 0.025 (0.254) FCF 0.003 0.028 -0.024 (-0.970) TANG 0.178 0.211 -0.033 (-0.640) SIZE 15.442 14.636 0.806* (1.895) AGE 8.914 8.202 0.712 (0.520) Panel E: Nigeria -0.006 -0.014 (-0.548) AAR -0.017 -0.006 -0.011 (-0.912) BTM 11.911 5.759 6.152** (2.262) SGR 0.213 0.297 -0.084 (-0.702) LQ 0.075 0.099 -0.024 (-0.969) LEV 0.199 0.177 0.022 (0.301) FCF -0.026 0.006 -0.032 (-0.803) TANG 0.314 0.313 0.001 (0.019) SIZE 17.539 16.569 0.970*** (3.280)	SGR	0.166	0.129	0.036	(0.565)
LEV0.2950.2700.025 (0.254) FCF0.0030.028 -0.024 (-0.970) TANG0.1780.211 -0.033 (-0.640) SIZE15.44214.6360.806* (1.895) AGE8.9148.2020.712 (0.520) Panel E: NigeriaROA0.0320.045 -0.014 (-0.548) AAR -0.017 -0.006 -0.011 (-0.912) BTM11.9115.759 6.152^{**} (2.262) SGR0.2130.297 -0.084 (-0.702) LQ0.0750.099 -0.024 (-0.969) LEV0.1990.1770.022 (0.301) FCF -0.026 0.006 -0.032 (-0.803) TANG0.3140.3130.001 (0019) SIZE17.53916.569 0.970^{***} (3.280) AGE2.2440.833 1.410^{***} (3.755)	LIQ	0.094	0.102	-0.008	(-0.241)
FCF0.0030.028-0.024(-0.970)TANG0.1780.211-0.033(-0.640)SIZE15.44214.6360.806*(1.895)AGE8.9148.2020.712(0.520)Panel E: NigeriaROA0.0320.045-0.014(-0.548)AAR-0.017-0.006-0.011(-0.912)BTM11.9115.7596.152**(2.262)SGR0.2130.0297-0.084(-0.702)LQ0.0750.099-0.024(-0.969)LEV0.1990.1770.022(0.301)FCF-0.0260.006-0.032(-0.803)TANG0.3140.3130.001(0.019)SIZE17.53916.5690.970***(3.280)AGE2.2440.8331.410***(3.755)	LEV	0.295	0.270	0.025	(0.254)
TANG 0.178 0.211 -0.033 (-0.640) SIZE 15.442 14.636 0.806* (1.895) AGE 8.914 8.202 0.712 (0.520) Panel E: Nigeria (0.520) (0.520) Panel E: Nigeria (0.520) (0.520) Panel E: Nigeria (0.520) (0.520) Panel AR -0.017 -0.006 -0.014 (-0.548) AAR -0.017 -0.006 -0.011 (-0.912) BTM 11.911 5.759 6.152** (2.262) SCR 0.213 0.297 -0.084 (-0.702) LIQ 0.075 0.099 -0.024 (-0.969) LEV 0.199 0.177 0.022 (0.301) FCF -0.026 0.006 -0.032 (-0.803) TANG 0.314 0.313 0.001 (0.019) SIZE 17.539 16.569 0.970*** (3.280) AGE	FCF	0.003	0.028	-0.024	(-0.970)
SIZE 15.442 14.636 0.806* (1.895) AGE 8.914 8.202 0.712 (0.520) Panel E: Nigeria ROA 0.032 0.045 -0.014 (-0.548) AAR -0.017 -0.006 -0.011 (-0.912) BTM 11.911 5.759 6.152** (2.262) SCR 0.213 0.297 -0.084 (-0.702) LQ 0.075 0.099 -0.024 (-0.969) LEV 0.199 0.177 0.022 (0.301) TANG 0.314 0.313 0.001 (0.019) SIZE 17.539 16.569 0.970*** (3.280) AGE 2.244 0.833 1.410*** (3.755)	TANG	0.178	0.211	-0.033	(-0.640)
ACE 8.914 8.202 0.712 (0.520) Panel E: Nigeria R0A 0.032 0.045 -0.014 (-0.548) AAR -0.017 -0.006 -0.011 (-0.912) BTM 11.911 5.759 6.152** (2.262) SCR 0.213 0.297 -0.084 (-0.702) LQ 0.075 0.099 -0.024 (-0.969) LEV 0.199 0.177 0.022 (0.301) FCF -0.026 0.006 -0.032 (-0.803) TANG 0.314 0.313 0.001 (0.019) SIZE 17.539 16.569 0.970*** (3.280) AGE 2.244 0.833 1.410*** (3.755)	SIZE	15.442	14.636	0.806*	(1.895)
Panel E: Nigeria ROA 0.032 0.045 -0.014 (-0.548) AAR -0.017 -0.006 -0.011 (-0.912) BTM 11.911 5.759 6.152** (2.262) SGR 0.213 0.297 -0.084 (-0.702) LQ 0.075 0.099 -0.024 (-0.969) LEV 0.199 0.177 0.022 (0.301) FCF -0.026 0.006 -0.032 (-0.803) TANG 0.314 0.313 0.001 (0019) SIZE 17.539 16.569 0.970*** (3.280) AGE 2.244 0.833 1.410*** (3.755)	AGE	8.914	8.202	0.712	(0.520)
Panel E: Nigeria ROA 0.032 0.045 -0.014 (-0.548) AAR -0.017 -0.006 -0.011 (-0.912) BTM 11.911 5.759 6.152** (2.262) SGR 0.213 0.297 -0.084 (-0.702) LQ 0.075 0.099 -0.024 (-0.699) EV 0.199 0.177 0.022 (0.301) FCF -0.026 0.006 -0.032 (-0.803) TANG 0.314 0.313 0.001 (0.019) SIZE 17.539 16.569 0.970*** (3.280) AGE 2.244 0.833 1.410*** (3.755)					
ROA 0.032 0.045 -0.014 (-0.548) AAR -0.017 -0.006 -0.011 (-0.912) BTM 11.911 5.759 6.152** (2.262) SGR 0.213 0.297 -0.084 (-0.702) LIQ 0.075 0.099 -0.024 (-0.969) EVV 0.199 0.177 0.022 (0.301) FCF -0.026 0.006 -0.032 (-0.803) TANG 0.314 0.313 0.001 (0.019) SIZE 17.539 16.569 0.970*** (3.280) AGE 2.244 0.833 1.410*** (3.755)	Panel E: Nigeria				
AAR -0.017 -0.006 -0.011 (-0.912) BTM11.9115.759 6.152^{**} (2.262) SGR0.2130.297 -0.084 (-0.702) LIQ0.0750.099 -0.024 (-0.969) LEV0.1990.1770.022 (0.301) FCF -0.026 0.006 -0.032 (-0.803) TANG0.3140.3130.001 (0.019) SIZE17.53916.569 0.970^{***} (3.280) AGE2.2440.8331.410^{***} (3.755)	ROA	0.032	0.045	-0.014	(-0.548)
BTM 11.911 5.759 6.152** (2.262) SCR 0.213 0.297 -0.084 (-0.702) LIQ 0.075 0.099 -0.024 (-0.969) LEV 0.199 0.177 0.022 (0.301) FCF -0.026 0.006 -0.032 (-0.803) TANG 0.314 0.313 0.001 (0.019) SIZE 17.539 16.569 0.970*** (3.280) AGE 2.244 0.833 1.410*** (3.755)	AAR	-0.017	-0.006	-0.011	(-0.912)
SGR 0.213 0.297 -0.084 (-0.702) LIQ 0.075 0.099 -0.024 (-0.969) LEV 0.199 0.177 0.022 (0.301) FCF -0.026 0.006 -0.032 (-0.803) TANG 0.314 0.313 0.001 (0.019) SIZE 17.539 16.569 0.970*** (3.280) AGE 2.244 0.833 1.410*** (3.755)	BTM	11.911	5.759	6.152**	(2.262)
LIQ 0.075 0.099 -0.024 (-0.969) LEV 0.199 0.177 0.022 (0.301) FCF -0.026 0.006 -0.032 (-0.803) TANG 0.314 0.313 0.001 (0.019) SIZE 17.539 16.569 0.970*** (3.280) AGE 2.244 0.833 1.410*** (3.755)	SGR	0.213	0.297	-0.084	(-0.702)
LEV 0.199 0.177 0.022 (0.301) FCF -0.026 0.006 -0.032 (-0.803) TANG 0.314 0.313 0.001 (0.019) SIZE 17.539 16.569 0.970*** (3.280) AGE 2.244 0.833 1.410*** (3.755)	LIQ	0.075	0.099	-0.024	(-0.969)
FCF -0.026 0.006 -0.032 (-0.803) TANG 0.314 0.313 0.001 (0.019) SIZE 17.539 16.569 0.970*** (3.280) AGE 2.244 0.833 1.410*** (3.755)	LEV	0.199	0.177	0.022	(0.301)
TANG 0.314 0.313 0.001 (0.019) SIZE 17.539 16.569 0.970*** (3.280) AGE 2.244 0.833 1.410*** (3.755)	FCF	-0.026	0.006	-0.032	(-0.803)
SIZE 17.539 16.569 0.970*** (3.280) AGE 2.244 0.833 1.410*** (3.755)	TANG	0.314	0.313	0.001	(0.019)
AGE 2.244 0.833 1.410*** (3.755)	SIZE	17.539	16.569	0.970***	(3.280)
	AGE	2.244	0.833	1.410***	(3.755)

Notes: Full variable definitions are given in the Table A2. ROA is the return on assets. AAR is the average monthly abnormal return computed using the market model. BTM is the ratio of book value of equity to market value of equity. SGR (sales growth) is the rate of change in total revenues from the previous period. LIQ (liquidity) is the ratio of cash and short term investments to total assets. LEV (leverage) is the debt to equity ratio. FCF is the ratio of free cash flow to total assets. TANG (tangible property) is the ratio of property, plant and equipment to total assets. SIZE is the natural log of total assets. AGE is the number of years since incorporation. NTargets refers to results for Non-targets. T statistics are shown in parenthesis. ***, **, * indicate significance at 1%, 5% and 10% levels, respectively.

The full sample results (Panel A) suggest that targets are more likely to be poorly performing (in terms of stock market returns), larger and older firms. On average, targets earn average monthly abnormal returns of -0.70% prior to receiving takeover bids compared -0.30% earned by non-targets. The difference is significant at the 10% level. We find that, on average, targets also generate lower accounting profits (ROA) when compared to non-targets. This result is consistent with the predictions of management inefficiency hypotheses as well as the empirical evidence from other markets (Powell and Yawson, 2007; Cremers et al., 2009; Brar et al., 2009). The average target has a size (Ln total assets) of 14.166 and age of 9.912 years compared to 13.816 and 7.722 years for non-targets, respectively. The difference in size and age is significant at the 1% level. This finding does not support prior evidence from other regions (e.g., UK, US, Europe) suggesting that targets are more likely to be smaller and younger than non-targets (Powell, 1997; Powell and Yawson, 2007; Cremers et al., 2009; Brar et al., 2009; Bhattacharjee et al., 2009). The finding suggests that the selection of African targets hinges on information asymmetry concerns as larger and more established firms (even in major economies such as South Africa, Egypt and Morocco, as shown in Panels B, C and D) are more likely to receive takeover bids. It also suggests that that affordability argument for the preference of smaller over big firms as potential targets (Palepu, 1986) might not be relevant in the African context. The univariate results for the other hypotheses do not reveal significant differences in the firm characteristics of African targets and non-targets. The full sample results are largely consistent with the results from subsample analysis (Panel B: South Africa, Panel C: Egypt, Panel D: Morocco, Panel E: Nigeria).

We further explore the results of the univariate analysis (Table 4) through logit regression analysis. Prior to running the model, we assess the likelihood of multicollinearity between our variables by computing Pearson and Spearman correlation coefficients, as well as, variance inflation factors (VIF) and tolerance values. The results from our multicollinearity diagnostics (untabulated) show a significant correlation between the independent variables. Nonetheless, the Pearson and Spearman correlation coefficients are quite low and are therefore unlikely to lead to issues of multicollinearity. The VIFs for all our variables are below the 3.0 threshold thus further eliminating any concerns of multicollinearity. These results are available upon request. In our logit models (Table 5), the dependent variable is a binary measure of takeover likelihood (it takes a value of 1 when a firm receives a takeover bid and zero, otherwise) and the independent variables the firm-level determinants of M&A activity.

In Table 5 we compare the characteristics of targets in the 5 years [Model (1)], 3 years [Model (2)], and 1 year [Model (3)] leading up to the bid to the characteristics of non-targets using a multivariate logit framework. In addition, we report results

Table 5
ogistic regression results for firm-level determinants of takeover likelihood.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
	5Year	3Year	1Year	Marginal Effects	Egypt	Morocco	Nigeria	South Africa
ROA	0.983***	0.795***	-0.277	-0.011	3.689	4.089	-11.575	0.731
	(0.000)	(0.001)	(0.130)	(0.152)	(0.132)	(0.514)	(0.149)	(0.221)
AAR	-2.206***	-3.646***	-3.118**	-0.129***	2.367	10.164**	-4.441	-2.428**
	(0.008)	(0.000)	(0.010)	(0.001)	(0.361)	(0.039)	(0.264)	(0.017)
BTM	0.002	-0.002	-0.011^{***}	-0.000^{***}	0.006	0.236**	-0.024	0.000
	(0.699)	(0.594)	(0.000)	(0.004)	(0.840)	(0.032)	(0.673)	(0.989)
SGR	0.160*	0.137***	0.156**	0.006***	0.077	-0.070	-0.309	0.023
	(0.061)	(0.008)	(0.011)	(0.000)	(0.730)	(0.943)	(0.805)	(0.829)
LIQ	0.347	0.547	0.503	0.021	-3.607^{**}	5.056*	-7.842	0.624
	(0.490)	(0.238)	(0.218)	(0.265)	(0.018)	(0.091)	(0.288)	(0.250)
LEV	0.005	0.015	-0.111^{*}	-0.005^{*}	0.409	-2.835	-2.169^{*}	-0.071
	(0.951)	(0.827)	(0.064)	(0.077)	(0.355)	(0.164)	(0.074)	(0.636)
TANG	-0.077	-0.071	0.005	0.000	-1.273	2.839	1.218	0.232
	(0.757)	(0.751)	(0.980)	(0.980)	(0.104)	(0.122)	(0.588)	(0.383)
FCF	-0.717^{*}	-0.792^{*}	-0.756^{**}	-0.031^{***}	-2.587	-3.221	2.480	-0.304
	(0.050)	(0.065)	(0.011)	(0.000)	(0.118)	(0.114)	(0.485)	(0.529)
SIZE	0.043	0.069*	0.085**	0.004	-0.004	-0.028	0.631	0.118***
	(0.356)	(0.092)	(0.041)	(0.108)	(0.970)	(0.908)	(0.327)	(0.002)
AGE	0.007	0.005	0.007	0.000	0.003	-0.088	-0.370^{**}	-0.015^{*}
	(0.594)	(0.669)	(0.561)	(0.527)	(0.953)	(0.321)	(0.018)	(0.100)
Constant	-2.290^{***}	- 3.095***	-4.346^{***}		-1.009	-3.191	-11.396	-2.932^{***}
	(0.000)	(0.000)	(0.000)		(0.531)	(0.313)	(0.272)	(0.000)
Obs	6306	6306	6306	6306	758	244	115	4471
Wald Chi ²	332.16***	1155.93***	772.04***	772.04***	19.09**	17.36*	36.41***	17.94*
Pseudo R ²	0.004	0.007	0.007	0.007	0.044	0.174	0.255	0.010

The dependent variable (BID) in the model is takeover likelihood which takes a value of 1 when a firm faces a takeover bid and a value of 0, otherwise. (1) to (4) use the full sample of firms while (5) to (8) focus on different subsamples. In (1) and (2), BID lags of 5 and 3 years, respectively, are imposed such that a BID is matched to a target's characteristics in the previous year as well as the last 5 (for (1)) and 3 (for (2)) years, respectively. Marginal effects in (4) are derived using the Delta method and refer to marginal effects for (3). Robust p-values in parentheses are adjusted for clustering at country level (15 clusters) in (1), (2) and (3) and for clustering at firm level in (5), (6), (7) and (8). ROA is the return on assets. AAR is the average monthly abnormal return computed using the market model. BTM is the ratio of book value of equity to market value of equity. SGR (sales growth) is the rate of change in total revenues from the previous period. LIQ (liquidity) is the ratio of cash and short term investments to total assets. LEV (leverage) is the firm's debt to equity ratio. FCF is the ratio of free cash flow (operating cash flow minus capital investments) to total assets. TANG (tangible property) is the ratio of property, plant and equipment to total assets. SIZE is the natural log of a firm's total assets. AGE is the number of years since incorporation. P-values are presented in parenthesis. ***, **, * indicate significance at 1%, 5% and 10% levels, respectively.

for marginal effects using the Delta method [Model (4)] as well as results for [Model (3)] based on different subsamples; Egypt [Model (5)], Morocco [Model (6)], Nigeria [Model (7)] and South Africa [Model (8)]. In essence, we use 5-, 3- and 1-year lags in models (1) to (3) to model the dynamics of target characteristics in the five-year period leading up to the takeover bid. In Models (5) to (8), we model takeover likelihood as a function of a firm's last observable characteristics (i.e., characteristics in the previous period). The results in Table 5 are based on clustered (by country and firm) robust standard errors. The results do not change qualitatively when we alternatively add country dummies, industry dummies or firm fixed effects. For brevity, we do not report alternative model specifications.

Targets appear to be firms that are profitable prior to the bid but lose profitability in the years leading up to the bid. We find that takeover likelihood increases with firm profitability in the five-year period leading up to the merger bid but declines with profitability in the bid year. Takeover likelihood also declines with monthly abnormal stock returns. These results support the management inefficiency hypothesis as it suggests that targets are, potentially, profitable firms which are inefficiently managed (i.e., experiencing decline in performance) in the period leading up to the merger bid. The results for Nigeria (insignificant) and South Africa (significant) provide some support to our finding that M&A plays a disciplinary role in African financial markets. Notwithstanding, we find that takeover likelihood in Morocco (significant) and Egypt (insignificant) increases with firm accounting and stock performance. The business environment in these North African countries (Middle East and North Africa or MENA region) is, perhaps, significantly different from the environment in Sub-Sahara Africa. Prior research on firms in the MENA region suggests that these firms are characterised by concentrated family ownership and blockholding (Bloom and van Reenen, 2007; Claessens et al., 2000). Nepotism and managerial entrenchment persists, as family members are typically selected to manage these corporations over long periods of time (Gedajlovic and Shapiro, 1998; Bloom and van Reenen, 2007; Claessens et al., 2000). As evident in our findings, the implication of this ownership and management structure is that the market for corporate control has a limited effect in this context. To a bidder, there are therefore no substantial benefits in merging with poorly performing firms as their managers cannot, generally, be easily replaced. This, perhaps, explains why well-performing firms make comparatively more suitable targets as bidders may directly benefit from the firm's continuing success without the need to replace incumbent management.

Our full sample evidence does not support the undervaluation argument (Palepu, 1986) as we do not find that takeover likelihood increases with potential firm undervaluation. One reason for this might be our earlier finding that African firms have very high book to market (BTM) ratios on average. Nonetheless, our results for Morocco shows that, consistent with the undervaluation hypothesis, potentially undervalued firms are more likely to receive takeover bids. Recall, that we find no evidence that M&A has a disciplinary role in this particular market. The role of M&A here is, perhaps, one of correcting market inefficiencies in firm valuation as opposed to managerial inefficiencies in managing firm performance.

Takeover likelihood increases with growth in sales (suggesting that targets possess potential for future growth or those in growth industry) but declines with firm leverage. The results from Models (1), (2) and (3) taken together suggest that targets reduce their levels of leverage in the years leading up to the merger bid. Targets also appear to increase their holdings of tangible assets in the period leading up to the merger bid. Consistent with the tangible property argument (Ambrose and

	(1) Firm-level		(2) Country-level		(3) Combined	
	Coef	(P-value)	Coef	(P-value)	Coef	(P-value)
ROA	-0.277	(0.130)			-0.752***	(0.003)
AAR	-3.118**	(0.010)			-3.168***	(0.000)
BTM	-0.011^{***}	(0.000)			-0.014^{***}	(0.000)
SGR	0.156**	(0.011)			0.157***	(0.000)
LIQ	0.503	(0.218)			0.301*	(0.066)
LEV	-0.111^{*}	(0.064)			-0.396***	(0.000)
TANG	0.005	(0.980)			0.095	(0.365)
FCF	-0.756^{**}	(0.011)			-0.353	(0.278)
SIZE	0.085**	(0.041)			0.149***	(0.000)
Age	0.007	(0.561)			-0.011^{***}	(0.000)
Comp1			0.092***	(0.009)	0.193***	(0.000)
Comp2			0.345***	(0.000)	0.214***	(0.000)
Comp3			-0.080	(0.291)	-0.107^{***}	(0.008)
Comp4			-0.065	(0.160)	0.023	(0.264)
Constant	-4.346^{***}	(0.000)	-4.373***	(0.000)	-6.243***	(0.000)
Obs	6306		7459		4244	

 Table 6

 Firm and country-level variables as predictors of takeover likelihood.

Notes: ROA is the return on assets. AAR is the average monthly abnormal return computed using the market model. BTM is the ratio of book value of equity to market value of equity. SGR (sales growth) is the rate of change in total revenues from the previous period. LIQ (liquidity) is the ratio of cash and short term investments to total assets. LEV (leverage) is the firm's debt to equity ratio. FCF is the ratio of free cash flow (operating cash flow minus capital investments) to total assets. TANG (tangible property) is the ratio of property, plant and equipment to total assets. SIZE is the natural log of a firm's total assets. AGE is the number of years since incorporation. P-values are presented in parenthesis. ****, **, * indicate significance at 1%, 5% and 10% levels, respectively.

Megginson, 1992), we find that takeover likelihood increases with the level of tangible assets held by firms. Nonetheless, the results are not significant at the 10% level. Inconsistent with the free cash flow argument (Jensen, 1986; Lehn and Poulsen, 1989; Powell, 1997), takeover probability declines with level of free cash flow. These results, when taken together with our finding on sales growth, suggests that takeover likelihood for African firms increases when firms have growth opportunities but lack free cash flow to exploit them. This suggests that takeovers in this region are motivated by the generation of synergies through the injection of financial resources.

Further, in support of the results from Table 4, our results from Table 5 show that takeover likelihood for African firms increases in firm size i.e., larger firms are more likely to receive takeover bids. As noted earlier, these results do not mirror the findings from other markets (UK, US, EU). In the African context, larger firms are more likely to be established entities with better corporate governance, more transparency, less information asymmetry and more stock market liquidity than their small counterparts making them more attractive as potential targets. Looking at the results from Models (1) to (3) together, there is some evidence, that a continuous increase in a firm's size (at least over the five year period) increases its visibility to bidders and its suitability as a potential target. The results of multivariate tests on firm age suggest that takeover probability generally increases with firm age across the sample. However, the regression coefficient of the firm age variable is not statistically significant at the 10% level. Subsample analysis shows a significant negative relation in the case of Nigeria and South Africa where younger firms are more likely to receive takeover bids.

5.4. Country- and firm-level factors as moderators of firm acquisition likelihood

We noted earlier that several prior studies explore country- and firm-level determinants on FDI and M&As in isolation. We argued this is, perhaps, a limited view as FDI through M&As will only be pursued under suitable country- and firm-level conditions. We next explore the extent to which firm- and country-level factors together explain a firm's takeover likelihood. Our design is to explore whether a model which combines both country- and firm-level factors has an incremental predictive ability over that which employees either. As shown in Table 6, we generate a firm-level takeover prediction model using only firm-level data, a country-level prediction model using only country-level data and a combined model which uses a combination of firm- and

Table 7 The explanatory power of firm-level, country-level and combined models.

	Obs.	Pseudo R2	AUC	Std. err.	Chi ² Diff. in AUC	
					Firm-level	Country-Level
Firm-level	4244	0.017	0.564	0.021		
Country-level	4244	0.140	0.596	0.018	1.410	
					(0.235)	
Combined	4244	0.152	0.639	0.019	22.030***	6.150**
(P-value)					(0.000)	(0.013)



Notes: This table compares the performance of firm-level, country-level and combined models. 'Firm-level' refers to model where firm characteristics are only predictors of takeover likelihood. 'Country-level' refers to model where country-level characteristics (national governance quality, economic performance and financial development) are the only predictors of acquisition likelihood. 'Combined' refers to a model combining both firm and country-level variables in determining firm takeover likelihood. The pseudo R square is the McKelvey and Zavoina's (1975) R squared. AUC is area under the Receiver Operating Characteristic (ROC) curve with its standard error given by Std. Err. The last two columns present results for difference in AUC between the three models. P-values for Chi² statistic are presented in parenthesis. ***, **, * indicate significance at 1%, 5% and 10% levels, respectively.

country-level data. We assess the performance of the three models using standard Pseudo R squares and Area under the Receiver Operating Curve (AUC) diagnostics (Table 7). AUC comparison is based on the non-parametric method discussed in DeLong et al. (1988). A model whose ROC curve equal to the diagonal line in the graphic (AUC = 0.50) has a predictive ability similar to a random guess. The bigger the differential between a model's ROC curve and the diagonal line (i.e., the larger its AUC), the higher is its predictive ability. A perfect model has an AUC of 1.

We find that, in this sample, our country-level determinants of M&A activity better explain firm takeover likelihood when compared to traditional firm-level factors. As shown in Table 7, the McKelvey and Zavoina's pseudo R square (and AUC) of the logit model that uses only country-level variables to assign firm takeover likelihood is 0.140 (0.596) and this is higher than the pseudo R square and AUC (0.017 and 0.564, respectively) of a model that uses only firm-level variables. Note that low pseudo R squares of this nature are typical in takeover likelihood modelling literature. For example, Powell and Yawson (2007) reports pseudo r squares of 0.02 (or 2%) for their main binomial logit model. The pseudo R square of the country-level model is over eight times that of the firm-level model. These results are gualitatively similar when other measures of pseudo R squares (including Cragg and Uhler's, Cox and Snell, McFadden's and Efron's R Squares) are employed. The difference in AUC between the two models is, nonetheless, not statistically significant at the 10% level (P-value of 0.235). A model which combines both firm-level and country-level factors as moderators of firm acquisition likelihood outperforms either models. The AUC achieved by the combined model (0.639) is also significantly higher than that achieved by either models. These results suggest that our country-level factors have significant implications for firm takeover likelihood. In fact, country-level factors which we advance in this study appear to better explain the incidence of takeovers amongst African firms when compared to traditional firm-level factors. Put differently, established merger prediction propositions (based on firm characteristics) do not fully explain the incidence of takeovers amongst listed African firms. This raises fresh questions about the value relevance of financial information produced by these firms as well as our understanding of what drives M&A in this context.

5.5. Additional analyses and sensitivity checks

Here, we conduct further robustness checks on some of our key results. First, in our analysis, we considered all bids irrespective of whether the bids were successful or eventually failed in resulting to a completed takeover. In untabulated results, we find that our conclusions do not materially change when we focus on the subset of completed or successful bids only (i.e., exclude failed bids). The results we obtain from this additional analysis are consistent with those in Tables 4 to 7. These results are available upon request.

Second, we explore the factors driving bids for control — those where the bidder aims to acquire more than 50% of the voting rights in the target. These acquisitions generally represent substantial investment in the target. In untabulated results, we find that this sub sample is characterised by two distinct features. First, although targets involved in such bids generally experience a decline in monthly abnormal returns prior to takeovers (as hypothesised), the likelihood of receiving a control bid in our African sample increases with firm accounting profitability. This suggests bidders are keen to gain control of firms with a track record and potential for profitability. Second, unlike other bids, the likelihood of receiving a control bid increases with firm free cash flow. Given that control bids are significant investments the presence of excess free cash flow in the target potentially allows the bidder to reduce some of the implicit acquisition costs. We did not restrict our sample to control bids in the main part of our study as this significantly reduces the number of bids in our sample.

For our third robustness check, we explore differences in the characteristics of targets of domestic and cross-border bids. We define domestic bids as bids from other African firms and cross-border bids as bids from outside the continent. In untabulated results, we find that the results for domestic bids are consistent with those presented in Table 5. Also consistent with the results in Table 5, the results for cross-border bids show that a decline in accounting profitability (management inefficiency) and a large firm size (information asymmetry concerns) are the main firm-specific factors explaining the selection of targets in the region by cross-border bidders.

Finally, Manne (1965) contends that the takeover market makes the corporate world a more efficient one by ensuring that managers who deviate from the best interest of their shareholders are replaced by more efficient management teams. A contentious issue which particularly relates to this region is whether the market for corporate control or takeover market as an external monitoring mechanism can enforce managerial discipline when the legal system is ineffective (Manne, 1965; Jensen and Ruback, 1983; Jensen, 1986). In untabulated results, we interact National governance variables with measures of management inefficiency in our logit analysis. We find some evidence that the market for corporate control works more efficiently (i.e., management inefficiency is associated with takeover bids) in the presence of strong national governance (as measured by Rule of Law index (ROLI) and Control of Corruption index (CCI)). That is, national governance and the market for corporate control are, perhaps, complements not substitutes in addressing management inefficiencies.

6. Summary and Conclusion

6.1. Summary of Findings

This study has investigated the antecedents of FDI through M&A for listed firms in African markets. Generally, the results support our contention that M&A volume across the 15 countries is moderated by location advantages prescribed in

Dunning's (1980) eclectic paradigm. Consistent with Dunning (1980), M&A volume increases with market size (the marketseeking motive of FDI: Markusen and Venables, 2000), human capital (the presence of knowledge and skilled labour: Noorbakhsh et al., 2001) and efficiency opportunities (low production cost opportunities: Asiedu and Lien, 2011). Inconsistent with eclectic paradigm (Dunning, 1980), but in line with the resource-curse paradox and prior empirical research on the subject (Sachs and Warner, 1995, 1997; Asiedu and Lien, 2011), we find a negative relation between M&A volume and resource endowments. We find evidence of a positive relation between a country's national governance quality and the volume of M&A in attracts. Better institutional environments (lower corruption, more effective governments, appropriate rule of law and government accountability) attract higher levels of FDI through M&A. Further, we find that the presence of an active stock market is a key ingredient in stimulating FDI through M&A. Countries with more developed stock markets (i.e., stock markets with several listed firms which are actively being traded) do attract more M&A. Overall, these results from country-level antecedents support our postulations that improvements in national governance, economic performance and stock market development over time, potentially, lead to improvements in the volume of M&A activity and the likelihood that firms will be acquired.

The results from firm-level antecedents suggest that M&A plays a key disciplinary role in many African financial markets, perhaps more efficient than the role of regulation given the challenges of enforcement in these markets. We find that the experience is heterogeneous across the continent with M&As in Morocco appearing to be more focused on correcting perceived market efficiencies in the valuation of firms than the inefficiencies in the management of firms. Further, in full sample and subsample (South Africa) analyses, we find evidence that M&A targets are more likely to be larger listed firms. This evidence is inconsistent with the traditional firm size hypothesis, but consistent with our argument that information asymmetry concerns underlie bidders' selection of suitable takeover targets in key African markets.

Our study is based on the premise that M&A activity fosters regional and national economic development by channelling resources to promising businesses, by correcting market inefficiency in the pricing of assets and through its ability to enforce managerial discipline, amongst others. Our results suggest that regulators have a pivotal role to play in the development of the market for corporate control as we find a strong association between national governance, economic conditions, stock market development and the volume of M&A activity, as well as, the likelihood that individual firms will receive takeover bids. Indeed, our evidence suggests that in this region, country-level factors are stronger determinants of firm acquisition likelihood than firm specific factors.

Our paper makes several unique contributions. The results show that time-varying country-level factors, such as the quality of national governance, economic performance and capital market development have a significant impact on M&A activity both in terms of the volume of activity and the likelihood that individual firms will be acquired. For the first time, we present large-sample cross-country evidence from Africa, which suggests that improvements in the quality of national governance, economic performance and stock market development are likely to lead to improvements in M&A activity. As expected, we also find that the likelihood that individual firms will receive takeover bids can be explained by factors beyond their characteristics i.e., country-level factors. Our results suggest that in this sample, these country-level factors better explain the incidence of takeovers when compared to traditional firm-level variables.

6.2. Limitations and areas for further research

This study focuses on M&A involving listed African companies. Arguably, a substantial number of firms in this region are unlisted. In recent years, a significant amount of FDI in Africa has hailed from China. The literature suggests that FDI decisions by Chinese acquirers are unlikely to be moderated by factors, such as location advantages and national governance quality explored in this study. Nonetheless, we are unable to explore this argument as, over our sample period, all but six Chinese acquisitions into Africa were for unlisted companies with limited data. Further, we have focused on key firm-level variables used in recent M&A studies (Powell and Yawson, 2007; Danbolt et al., forthcoming) and, due to data unavailability, have ignored variables linked to corporate governance (such as board size & board independence, CEO-chairman duality), which might also impact on the likelihood for some firms to be involved in M&As.

This study opens up opportunities for future research in the area. Our results suggest that the M&A experience is heterogeneous across different African countries. For example, we find that while poor stock market performance increases takeover likelihood for firms in South Africa (as expected), the reverse is true for firms in Morocco. Future studies can further explore the contextual factors that shape the market for corporate control across these different countries. There are opportunities to expand the country-level determinants of African M&As by looking at variables related to inter-country trade. Clearly, substantial international trade (exports, imports) between two countries can foster business ties and facilitate M&As between these countries. Further, Our results show that firm-level variables and established prediction hypothesis (Palepu, 1986; Ambrose and Megginson, 1992; Powell and Yawson, 2007; Danbolt et al., forthcoming) do not really explain firm takeover likelihood in the context of Africa. This raises new questions about the value-relevance of financial information generated by these firms, as well as the transferability of these established theories to a unique sample such as the African market. Future research can explore the extent to which IFRS adoption by African countries facilitates or stimulates M&A activity and also how other firm, industry and market characteristics impact on bidders' acquisition decisions in this context. Overall, this study extends the scant literature on M&As in African markets and contributes to our understanding of factors shaping the bidders' choice of suitable takeover targets.

Appendix A

Table A1

Country-level determinants of takeover likelihood.

Determinants	Proxies (Exp. sign)	Variable definition
Market size	LnGDP (+)	Natural log of GDP. Source; World Bank Group
	GDPGrowth (+)	Growth in GDP. Source; World Bank Group
Resource Endowments	ResourceRent	Total natural resources rents (as a percentage of GDP). Source; World Bank Group
Human Capital	Patent	Total number of patent applications made by residents. Source; World Bank Group
Efficiency opportunities	FuelPrice	Dollar price per litre of diesel. Source; World Bank Group
	AverageWage	Average wage in dollars. Source; World Bank Group
National governance quality	CPI (+)	Corruption Perception Index. Source; Transparency International
	GEI (+)	Government Effectiveness Index. Source; WGI
	VAI (+)	Voice and Accountability Index. Source; WGI
	RQI (+)	Regulatory Quality Index. Source; WGI
	ROLI (+)	Rule of Law Index. Source; WGI
	CCI (+)	Control of Corruption Index. Source; WGI
Stock market	SMCap (+)	Stock market capitalisation. Source; World Bank Group
development	MktVol (+)	Stock market volatility. Source; World Bank Group
	ValueTraded (+)	Stock market value traded as a proportion of market capitalisation. Source;
		World Bank Group
	TradedStocks (+)	Number of stocks traded as a proportion of total listed firms in Africa

Notes: Worldwide Governance Indicators (WGI) are compiled by Kaufmann et al. (2007, 2010) and freely available on the World Bank Group website (Worldbank.org). The methodology for their development is discussed in Kaufmann et al. (2010). Data for country-level Economic and Financial development indicators is compiled by the World Bank Group.

Table A2

Firm-level determinants of takeover likelihood.

Determinants	Proxies (Exp. sign)	Variable definition
Inefficient management	ROA (-) AAR (-)	Return on assets: Net income to total assets ratio Average excess stock returns over the last 12 months, measured against the Morgan Stanley Capital International (MSCI) Emerging Markets Index
Undervaluation	BTM (+)	Ratio of total assets less intangibles to market value
Free cash flow	FCF(+)	Ratio of operating cash flow less capital expenditures to total assets
Growth-resource	SGR (+/-)	Sales growth: Percentage change in sales
Mismatch	LIQ (+/-)	Liquidity: Cash and equivalents to total assets ratio
	LEV (+/-)	Leverage: Long term debt to equity ratio
Tangible assets	TANG (+)	Tangibility: Property, plant and equipment to total assets
Firm size	SIZE (+)	Natural log of total assets
Firm age	AGE (-)	Number of days since incorporation/365

Notes: Firm-level variables used to develop proxies for the hypotheses are obtained from Thomson DataStream database. The proxies for these hypotheses, together with their expected signs, are shown in the second column.

References

Agarwal, R., Gort, M., 2002. Firm and product life cycles and firm survival. Am. Econ. Rev. 92 (2), 184–190.

Agrawal, A., Jaffe, J.F., 2003. Do takeover targets underperform? Evidence from operating and stock returns. J. Financ. Quant. Anal. 38 (4), 721-746.

Akerlof, G., 1970. The market for "lemons": quality uncertainty and the market mechanism. Q. J. Econ. 84 (3), 488–500.

Allen, F., Carletti, E., Cull, R., Qian, J., Senbet, L., Valenzuela, P., 2012. Resolving the African financial development gap: cross-country comparisons and a within-country study of Kenya. In: Edwards, S., Johnson, S., Weil, D. (Eds.), Forthcoming in NBER Volume on African Economic Successes. University of Chicago Press.

Alvarez, I., Marin, R., 2010. Entry modes and national systems of innovation. J. Int. Manag. 16 (4), 340–353.

Ambrose, B.W., Megginson, W.L., 1992. The role of asset structure, ownership structure, and takeover defences in determining acquisition likelihood. J. Financ. Quant. Anal. 27 (4), 575–589.

Andrade, G., Stafford, E., 2004. Investigating the economic role of mergers. J. Corp. Finance 10 (1), 1–36.

Andrade, G., Mitchell, M., Stafford, E., 2001. New evidence and perspectives on mergers. J. Econ. Perspect. 15 (2), 103–120.

Andrianaivo, M., Yartey, C.A., 2010. Understanding the growth of African financial markets. Afr. Dev. Rev. 22 (3), 394-418.

Asiedu, E., Lien, D., 2011. Democracy, foreign direct investment and natural resources. J. Int. Econ. 84, 99–111.

Bhattacharjee, A., Higson, C., Holly, S., Kattuman, P., 2009. Macroeconomic instability and business exit: determinants of failures and acquisitions of UK firms. Economica 76 (301), 108–131.

Bloom, N., Van Reenen, J., 2007. Measuring and explaining management practices across firms and countries. Q. J. Econ. 122 (4), 1351–1408. Boas, M., 1998. Governance as multilateral Development Banks policy: the cases of the African Development Bank and the Asian Development Bank. Eur. J. Dev. Res. 10 (2), 117 - 134.Brainard, S.L., 1997. An empirical assessment of the proximity-concentration trade-off between multinational sales and trade. Am. Econ. Rev. 87, 520-544. Brar, G., Giamouridis, D., Liodakis, M., 2009. Predicting European takeover targets. Eur. Financ. Manage. 15 (2), 430-450. Cerny, C.A., Kaiser, H.F., 1977. A study of a measure of sampling adequacy for factor-analytic correlation matrices. Multivar. Behav. Res. 12 (1), 43–47. Claessens, S., Djankov, S., Lang, L.H., 2000. The separation of ownership and control in East Asian corporations. J. Financ. Econ. 58 (1), 81-112. Coase, R., 1937. The nature of the firm. Economica 4 (16), 386-405. Cremers, K.M., Nair, V.B., John, K., 2009. Takeovers and the cross-section of returns. Rev. Financ. Stud. 22 (4), 1409–1445. Danbolt, J., Siganos, A., Tunyi, A., 2016. Abnormal returns from takeover prediction modelling: challenges and suggested investment strategies. J. Bus. Finan. Acc. (forthcoming). DeLong, E.R., DeLong, D.M., Clarke-Pearson, D.L., 1988. Comparing the areas under two or more correlated receiver operating characteristic curves: a nonparametric approach. Biometrics 44 (3), 837-845. Dikova, D., van Witteloostuijn, A., 2007. Foreign direct investment mode choice: entry and establishment modes in transition economies. J. Int. Bus. Stud. 38, 1013-1033. Dunning, J.H., 1980. Toward an eclectic theory of international production: some empirical tests. J. Int. Bus. Stud. 11 (1), 9-31. Dunning, J.H., 1998. Location and the multinational enterprise: a neglected factor. J. Int. Bus. Stud. 29 (1), 45-66. Dunning, J.H., 2009. Location and the multinational enterprise; a neglected factor? J. Int. Bus. Stud. 40, 5–19. Eleswarapu, V., Thompson, R., Venkataraman, K., 2004. The impact of regulation fair disclosure: trading costs and information asymmetry. J. Financ. Quant. Anal. 39 (2), 208-225. Espahbodi, H., Espahbodi, P., 2003. Binary choice models and corporate takeover. J. Bank. Financ. 27 (4), 549-574. Ferraz, J., Hamaguchi, N., 2002. Introduction: M&A and privatization in developing countries. Dev. Econ. 40 (4), 383-399. Franco, C., Rentocchini, F., Marzetti, G., 2010. Why do firms invest abroad? An analysis of the motives underlying foreign direct investments. IUP J. Int. Bus. Law 9 (182), 42-65. Gedajlovic, E.R., Shapiro, D.M., 1998. Management and ownership effects: evidence from five countries. Strateg. Manag. J. 19 (6), 533-553. Gomes, E., Angwin, D., Peter, E., Mellahi, K., 2012. HRM issues and outcomes in African mergers and acquisitions: a study of the Nigerian banking sector. Int. J. Hum. Resour. Manag. 23 (14), 2874–2900. Gorton, G., Kahl, M., Rosen, R., 2009. Eat or be eaten: A theory of mergers and firm size. J. Financ. 64 (3), 1291–1344. Guisinger, S., 2001. From OLI to OLMA: incorporating higher levels of environmental and structural complexity into the eclectic paradigm. Int. J. Econ. Bus. 8, 257–272. Harford, J., 2005. What drives merger waves? J. Financ. Econ. 77 (3), 529-560. Hearn, B., Piesse, J., Strange, R., 2010. Market liquidity and stock size premia in emerging financial markets: the implications for foreign investment. Int. Bus. Rev. 19 (5), 489-501 Hopenhayn, H.A., 1992. Entry, exit, and firm dynamics in long run equilibrium. Econometrica 60 (5), 1127–1150. Hymer, S.H., 1960a. The International Operations of National Firms: A Study of Direct Foreign Investment (PhD Thesis) Massachusetts Institute of Technology. Hymer, S.H., 1960b. The International Operations of National Firms: A Study of Direct Foreign Investment. MIT Press, Cambridge. Hymer, S.H., 1968. The large multinational corporation: an analysis of some motives for the international integration of business. Rev. Econ. 19 (6), 949–973. Jensen, M.C., 1986. Agency costs of free cash flow, corporate finance, and takeovers. Am. Econ. Rev. 76 (2), 323-329. Jensen, M.C., Ruback, R.S., 1983. The market for corporate control: the scientific evidence. J. Financ. Econ. 11 (1-4), 5-50. Karhunen, P., Ledyaeva, S., 2012. Corruption distance, anti-corruption laws and international ownership strategies. J. Int. Manag. 18, 196-208. Kaufmann, D., Kraay, A., Mastruzzi, M., 2007. Governance matters VI: governance indicators of 1996–2006. World Bank Research Working Paper 4280 (Washington, DC). Kaufmann, D., Kraay, A., Mastruzzi, M., 2010. The worldwide governance indicators: methodology and Analytical Issue. World Bank Research Working Paper 5430 (Washington, DC) Lasserre, P., 1996. Regional headquarters: the spearhead for Asia Pacific markets. Long Range Plan. 29 (1), 30–37. Lehn, K., Poulsen, A., 1989. Free cash flow and stockholder gains in going private transactions. J. Financ, 44 (3), 771–787. Li, D., Miller, S., Eden, L., Hitt, M., 2012. The impact of rule of law on market value creation for local alliance partners in BRIC countries. [. Int. Manag. 18 (4), 305–321. Luiz, J., Radebe, B., 2012. The strategic location of regional headquarters for multinationals in Africa. University of Witwatersrand Working Paper No. 276. Maksimovic, V., Phillips, G., 2001. The market for corporate assets: who engages in mergers and asset sales and are there efficiency gains? J. Financ. 56 (6), 2019–2065. Manne, H.G., 1965. Mergers and the market for corporate control. J. Polit. Econ. 73 (2), 110-120. Markusen, J.R., 1984. Multinationals, multi-plant economies, and the gains from trade. J. Int. Econ. 16 (3-4), 205-226. Markusen, J.R., Venables, A.J., 1998. Multinational firms and new trade theory. J. Int. Econ. 46, 183–203. Markusen, J.R., Venables, A.J., 2000. The theory of endowment, intra-industry and multinational trade. J. Int. Econ. 52, 209–234. Martynova, M., Renneboog, L., 2008. A century of corporate takeovers: what have we learned and where do we stand? J. Bank. Financ. 32 (10), 2148–2177. McFerson, H., 2009. Hypercorruption in resource-rich African countries. Third World Q. 30 (8), 1529–1548. McFerson, H., 2010. Developments in African governance since the cold war: beyond Cassandra and Pollyanna. Afr. Stud. Rev. 53 (2), 49–76. McKelvey, R.D., Zavoina, W., 1975. A statistical model for the analysis of ordinal level dependent variables. J. Math. Sociol. 4, 103–120. Mitchell, M.L., Mulherin, J.H., 1996. The impact of industry shocks on takeover and restructuring activity. J. Financ. Econ. 41 (2), 193-229. Moghaddam, K., Sethi, D., Weber, T., Wu, J., 2014. The smirk of emerging market firms: a modification of the Dunning's typology of internationalization motivations. J. Int. Manag. 20 (3), 359-374. Morck, R., Shleifer, A., Vishny, R., 1989. Alternative mechanisms for corporate control. Am. Econ. Rev. 79 (4), 842–852. Mueller, D.C., 1969. A theory of conglomerate mergers. Q. J. Econ. 83 (4), 643-659. Narula, R., 1996. Multinational Investment and Economic Structure: Globalisation and Competitiveness. Routledge, London. Noorbakhsh, F., Paloni, A., Youssef, A., 2001. Human Capital and FDI Inflows to Developing Countries: New Empirical Evidence. World Dev. 26 (7), 1593-1610. North, D.C., 2005. Introduction to understanding the process of economic change. Washington University Press, Washington, USA. Ntim, C.G., Opong, K.K., Danbolt, J., 2007. An empirical re-examination of the weak form efficient markets hypothesis of the Ghana stock market using variance-ratios tests. Afr. Financ. J. 9 (2), 1-25. Ntim, C.G., Opong, K.K., Danbolt, J., Dewotor, F.S., 2011. Testing the weak-form efficiency in African stock markets. Manag. Financ. 37 (3), 195–218. Ntim, C.G., 2012. Why should African stock markets harmonise and integrate their operations. Afr. Rev. Econ. Finance 4 (1), 53–72. Ntim, C.G., Opong, K.K., Danbolt, J., 2012. The relative value relevance of shareholder versus stakeholder corporate governance disclosure policy reforms in South Africa. Corp. Gov. 20, 84-105. Ntim, C.G., English, J., Nwachukwu, J., Wang, Y., 2015. On the efficiency of the global gold markets. Int. Rev. Financ. Anal. 41, 218–236. OECD, 2006. OECD Investment Policy Reviews: China Open Policy Towards Mergers and Acquisitions. OECD Publications.

Pakes, A., Ericson, R., 1998. Empirical implications of alternative models of firm dynamics. J. Econ. Theory 79 (1), 1-45.

Palepu, K., 1986. Predicting takeover targets: a methodological and empirical analysis. J. Account. Econ. 8, 3–36.

Pettit, R., Singer, R., 1985. Small business finance: a research agenda. Financ. Manag. 13 (3), 47-60.

Pfeffermann, G., Madarassy, A., 1992. Trends in private investment in developing countries, 1992 edition. International Finance Corporation discussion paper; No. 14, IFD/IFC working paper series. the World Bank Working Paper, Washington, DC.

Poelhekke, S., van der Ploeg, F., 2010. Do natural resources attract FDI? Evidence from nonstationary sector level data. DNB Working Paper No. 266/November 2010, Germany.

Powell, R.G., 1997. Modelling takeover likelihood. J. Bus. Financ. Acc. 24 (7-8), 1009-1030.

Powell, R.G., 2001. Takeover prediction and portfolio performance: a note. J. Bus. Financ. Acc. 28 (7–8), 93–1011.

- Powell, R.G., 2004. Takeover prediction models and portfolio strategies: a multinomial approach. Multinat. Financ. J. 8 (1), 35–74.
- Powell, R.G., Yawson, A., 2007. Are corporate restructuring events driven by common factors? Implications for takeover prediction. J. Bus. Financ. Acc. 34 (7–8), 1169–1192.
- Rhodes-Kropft, M., Viswanathan, S., 2004. Market valuation and merger waves. J. Financ. 59 (6), 2685–2718.
- Rogers, W., 1993. Regression standard errors in clustered samples. Stata Tech. Bull. 13, 19–23.
- Root, F., Ahmed, A., 1979. Empirical determinants of manufacturing direct foreign investment in developing countries: a case study of Taiwan. Weltwirtschaftliches Arch. 111, 505–528.
- Rossi, S., Volpin, P., 2004. Cross-country determinants of mergers and acquisitions. J. Financ. Econ. 74, 277–304.
- Sachs, J.D., Warner, A.M., 1995. Natural resource abundance and economic growth. NBER Working Paper Series, 5398, pp. 1-47.
- Sachs, J., Warner, A., 1997. Natural Resource and Economic Growth. NBER Working Paper No. 5398.
- Schneider, F., Frey, B., 1985. Economic and political determinants of foreign direct investment. World Dev. 13, 41-51.
- Shimizu, K., Hitt., M., Vaidyanath, D., Pisano, V., 2004. Theoretical foundations of cross-border mergers and acquisitions: a review of current research and recommendations for the future. J. Int. Manag. 10 (3), 307–353.
- Stoian, C., Filippaios, F., 2008. Dunning's eclectic paradigm: a holistic, yet context specific framework for analysing the determinants of FDI; evidence from International Greek Investments. Int. Bus. Rev. 17 (3), 349–367.
- Trautwein, F., 1990. Merger motives and merger prescriptions. Strateg. Manag. J. 11 (4), 283–295.
- Vencatachellum, D., Wilson, M., 2013. Determinants of M&As targeting Africa 1990–2011. African Development Bank Working Paper Available at: http://www.afdb. org/en/aec-2013/papers/paper/determinants-of-mergers-and-acquisitions-targeting-africa-1990-2011-1027/ (Accessed on 30 March 2015).
- Voyer, P., Beamish, P., 2004. The effect of corruption on Japanese foreign direct investment. J. Bus. Ethics 50 (3), 211–224.
- Wang, Y., Xiaoyu, L., Abdou, H., Ntim, C., 2015. Financial development and economic growth in China. Invest. manag. Financ. Innov. 12 (3), 8–18.
- Wei, S., 2000. Local corruption and global capital flows. Brook. Pap. Econ. Act. 2, 303–346.
- Zhao, J., Kim, S., Du, J., 2003. The impact of corruption and transparency on foreign direct investment: an empirical analysis. Manag. Int. Rev. 43 (1), 41–62.